

Temi di discussione

(Working Papers)

Fiscal space and the size of fiscal multiplier

by Luca Metelli and Kevin Pallara







Temi di discussione

(Working Papers)

Fiscal space and the size of fiscal multiplier

by Luca Metelli and Kevin Pallara

Number 1293 - September 2020

The papers published in the Temi di discussione *series describe preliminary results and are made available to the public to encourage discussion and elicit comments.*

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Federico Cingano, Marianna Riggi, Monica Andini, Audinga Baltrunaite, Marco Bottone, Davide Delle Monache, Sara Formai, Francesco Franceschi, Salvatore Lo Bello, Juho Taneli Makinen, Luca Metelli, Mario Pietrunti, Marco Savegnago. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 1594-7939 (print) ISSN 2281-3950 (online)

Printed by the Printing and Publishing Division of the Bank of Italy

FISCAL SPACE AND THE SIZE OF THE FISCAL MULTIPLIER

by Luca Metelli^{*} and Kevin Pallara[^]

Abstract

This paper investigates the interaction between fiscal policy transmission and fiscal sustainability, captured through the concept of fiscal space. In order to measure the evolution of fiscal space over time we propose four indicators, drawing from different concepts available in the literature. We use these indicators to define periods of ample and tight fiscal space. We then estimate the effects of government spending shocks in the United States according to the level of fiscal space, for the period 1929:Q1-2015:Q4. The main finding of the paper is that the fiscal multiplier is above one when fiscal space is ample, while it is below one when fiscal space is tight. Moreover, such difference is always significant. This result is very robust across different identification methods and samples.

JEL Classification: E62, H50, H60.

Keywords: fiscal multipliers, fiscal space, state-dependency, local projection, fiscal sustainability, government spending. **DOI**: 10.32057/0.TD.2020.1293

Contents

1. Introduction	5
2. Literature review	9
3. Fiscal space approximations	13
4. Methodology	
5. Results	
6. Conclusions	
Figures and tables	
References	
Appendices	54

^{*} Bank of Italy, Directorate General for Economics, Statistics and Research - International Relations and Economic Research.

[^] University of Lausanne, Department of Economics.

1. INTRODUCTION¹

Since the Great Recession, the debate on the role of fiscal policy has gained traction, as discretionary fiscal measures have started afresh to serve as policy tools in advanced economies. Large spending plans have been implemented in many advanced economies and especially in the United States. However, growing deficits piled up into unprecedented levels of public debt. The latter, together with stagnant growth and low inflation, raised the attention on the sustainability of public finances and called into question whether the effects of fiscal policy as well might depend on fiscal sustainability considerations. According to this view, fiscal policy can prove to be a powerful tool in certain situations while not in others. In particular, an expansion in the public budget associated with a weak fiscal position can even produce harmful effects, while, at the opposite, the same fiscal shock implemented when public finances are sound generates expansionary effects. In this paper we investigate such hypothesis empirically. Addressing this conjecture is key also at the light of the ongoing situation generated by Covid-19. Governments around the world are expanding massively their budget deficits, in an effort to mitigate the detrimental consequences of the pandemic. While these policies are certainly necessary to face the emergency, they will generate a strong deterioration in public finance sustainability in the medium-term, which might affect the effectiveness of future fiscal policy shocks. Indeed, seminal contribution from Perotti (1999) already pointed out that shocks to government expenditure in times of fiscal stress have very different effects on the economy than in normal times.²

This empirical work studies whether the transmission mechanism of fiscal policy is affected by the state of fiscal sustainability. In order to take into account fiscal sustainability we refer to the notion of fiscal space. Heller (2005) provides the following definition for fiscal space:³

¹We are grateful to Marco Bernardini, Florin Bilbiie, Auréliene Eyquem, Valerio Ercolani, Andrea Finicelli, Andrea Gazzani, Jean-Paul Renne, Paolo Surico, Alex Tagliabracci and Andreas Tischbirek for interesting discussions and comments on preliminary ver-sions of this work. We are also grateful to seminar participants at Banca d'Italia, HEC University of Lausanne and Monash University (Applied Young Economists webinar) and to participants at the SIdE-IEA Workshop and the Workshop in Structural VAR models (3rd edition), especially to Luca Fanelli and Regis Barnichon for their discussion.

²Previous studies (Blanchard, 1990; Sutherland, 1997) showed that fiscal consolidation in the form of tax increases lead to non-Keynesian effects in times of weak fiscal position (namely, high debt or deficits).

³A similar definition can be found in Ley (2009) and Escolano (2010).

Room in a government's budget that allows it to provide resources for a desired purpose without jeopardizing the sustainability of its financial position or the stability of the economy.

However, there is no agreement on how to translate such a loose notion into a proper measure. In the definition provided above, the link to the concept of fiscal sustainability is explicit. This relates to the ability of the government to fund its preferred spending programs, while being able to service its obligations and to ensure solvency. On the contrary, Bi (2012), Bi and Leeper (2013) and Ghosh et al. (2013b) regard fiscal space as the distance between the level of current debt-to-GDP ratio and a country specific debt limit, which represents the maximum amount of debt that an economy can credibly sustain. Additionally, Perotti (2007) delineates the concept of fiscal space as a different approach in setting up the intertemporal budget constraint. The key characteristic of fiscal space is to incorporate considerations on the ability of the government to service its debt, which relates, in turn, to the dynamics of macroeconomic variables like the interest rate together with perceived sovereign risk, the GDP growth rate, the amount of primary surplus and the ratio of debt-to-GDP. In front of these multiple interpretations regarding fiscal space, we propose different methods to track its evolution over time, each relating to different underlying theories, as we clarify later.

While the literature (e.g. Huidrom et al., 2019; Ilzetzki et al., 2013; Auerbach and Gorodnichenko, 2017) focused on debt-to-GDP ratio to capture the role of fiscal sustainability in the transmission of fiscal policy, we take a more comprehensive approach through the concept of fiscal space. Fiscal space is far from being a static concept and cannot be determined only by a country's public debt level. Although correlated with debt-to-GDP, fiscal space encompasses other crucial aspects. First of all, it measures the overall ability of the government to service its obligations, which depends only in part on the level of public debt to be repaid. Second, it considers public finance aggregates jointly with other key macroeconomic variables, taking also into account the fundamental debt capacity of the economy. Lastly, the forward-looking nature of fiscal space contrasts with the path-dependent nature of a stock variable like debtto-GDP. Indeed fiscal space varies with market and economic conditions, which often change abruptly. For instance, a productive fiscal stimulus could improve the economic outlook in a country such that fiscal space improves while increasing its public debt-to-GDP ratio in the short run or, on the contrary, a harmful fiscal tightening could ameliorate a country's indebtedness while worsening growth and, thus, reducing the perceived fiscal room. Therefore, we avoid focusing on a single metric, like the debt-to-GDP ratio, and we rely on a multi-faceted approach using more indicators based on different methods within the relevant literature to measure the dynamic concept of fiscal space.

The idea of a differential effect of fiscal policy according to the fiscal position fits in the more general debate on fiscal policy, which has established a consensus on the fact that there is not such thing as a unique fiscal multiplier, but, more likely, the effects of fiscal shocks are state dependent. This literature, however, focused mainly on studying how fiscal policy's effects vary with the business cycle, differentiating in particular recession versus expansion periods, and with the monetary policy stance, with a particular reference to periods in which this is constrained by the zero lower bound. By contrast, there are no studies considering fiscal sustainability as a state variable in the transmission of fiscal policy. In this paper we investigate this form of state dependency and we aim at answering the following questions. How can we measure the evolution of fiscal space over time? Do the effects of fiscal shocks depend upon the level of fiscal space? If so, what is the rationale behind such differential effect?

The paper addresses the aforementioned questions in the following way. In the first part, we build four different indicators to measure the evolution of fiscal space over time, using data for the US. Our preferred indicator relates to the concept of primary surplus sustainability gap as in Kose et al. (2017) and we calculate, at each point in time, the primary surplus needed to stabilize the trajectory of public debt. Such simple yet effective indicator captures well periods of high debt velocity and inherent inability to contain debt roll-over needs via primary surpluses. The second indicator draws from the theoretical literature on fiscal limits as in Bi (2012) and Bi and Leeper (2013) and represents its empirical counterpart. The third one related to the idea of fiscal imbalances of Auerbach (1997), while the last one builds on Aizenman and Jinjarak (2010). We refer to Section 3 for further details on the computation of these indicators. After the construction of such fiscal space indicators, we show that they correlate well among each other, reinforcing the idea that, although derived from different underlying theories, they capture slightly different aspects of the same phenomenon we want to measure, i.e. the evolution over time of fiscal sustainability. We also provide evidence that our measures do not confound with other cyclical indicators, like the economic cycle or ZLB periods. In the second part of the paper, we estimate empirically the effects of fiscal policy in a state-dependent fashion, differentiating periods in which fiscal space is tight and when fiscal space is ample. We estimate the effects of government spending shocks in the US for the period 1929-2015, using two different identification methods, the one of Ramey (2011b,a) and the one of Blanchard and Perotti (2002).⁴ We then employ the Local-Projections method developed by Jordà (2005) to estimate the state-dependent effect of fiscal policy, using our four indicators to define, for each proxy, a tight fiscal space state when the proxy is above the median and a large fiscal space state when it is below. We quantify the impact of fiscal policy in the large and tight fiscal space state calculating the cumulative fiscal multiplier, as in Ramey and Zubairy (2018). We also investigate the mechanism behind our results, analyzing the effects on other variables other than output, like private consumption, private investment, interest rates and debt-to-GDP ratio. In order to quantify what transmission channel is more relevant, we calculate multipliers also for consumption and investment.

The main results of the paper are the following. First, we find that fiscal policy is much more effective when implemented in periods associated with large fiscal space. The corresponding fiscal multiplier is above one, while by contrast, when fiscal space is tight the fiscal multiplier is smaller than one, with a difference in the two cases always statistically significant. This result is particularly relevant in light of the findings by other studies on non-linearities in fiscal policy. While Ramey and Zubairy (2018) conclude there is no difference in the effects of fiscal policy between expansion and recession and find only minor dissimilarities when monetary policy is at the ZLB, our paper identifies a major distinction in the size of fiscal multiplier across different fiscal space regimes, suggesting that such nonlinearity might be the relevant one. Second, we show that our result occurs independently of the identification method adopted and is robust across different samples and different empirical specifications. More importantly, the result is strikingly similar across the four fiscal space indicators we construct, signaling that our indicators capture the same phenomenon. We also implement the estimations using debt-to-GDP ratio as the state variable. In this case we do not find difference in the two states, suggesting the importance of looking at specific indicators of fiscal space, in contrast to other variables, when studying fiscal sustainability

⁴We use such an historical perspective as we use Ramey news to instrument for government spending, which can be done only considering a long enough sample.

issues. Finally, the paper represents a first step to investigate the rationale behind the differential response of fiscal policy in the two s tates. We show that private consumption and investment follow a very different behaviour across states. Indeed, in case of ample fiscal space, government spending shock does not generate the standard Ricardian effect as in the case of weak fiscal space but, on the contrary, produces an increase in private consumption and does not crowd out investment.

This paper contributes to the literature in the following ways. First, we provide an historical time series for fiscal space in the US starting from 1929 up to 2015, at the quarterly frequency, according to four different indicators. The only database containing time-varying fiscal space measures, Kose et al. (2017), is at annual frequency and starts from 2001. We extend some of the measures present in Kose et al. (2017) backwards and we provide other fiscal space proxies drawing from multiple notions and documenting the variability over time of fiscal space in the US. Second, this is the first paper to investigate empirically the effects of fiscal policy according to fiscal space conditions. While Huidrom et al. (2019), Ilzetzki et al. (2013), Auerbach and Gorodnichenko (2017) estimated fiscal multipliers according to different levels of public debt, we do so using fiscal space. Finally, we also delve into an investigation of the transmission mechanism, highlighting the striking difference in the response of private consumption and investment across states of fiscal space.

The rest of the paper is organized as follows. Section 2 provides a literature review, while Section 3 describes carefully the procedure we follow to construct our fiscal space indicators and their properties. Section 4 provides details on the empirical methodology, the data and the identification method a dopted. Section 5 presents the empirical results, together with a robustness section. Finally, Section 6 concludes.

2. LITERATURE REVIEW

Our paper is related to different strands of the literature. First of all, we relate to the literature, both empirical and theoretical, aiming at investigating and measuring the concept of fiscal s pace. Part of this literature provides model-free estimates of fiscal space proxies. Given that fiscal space is a multi-dimensional concept, Kose et al. (2017) propose an extensive dataset collecting model-free proxies for fiscal space for several countries. The paper collects a large cross-sectional database including 28 indicators, which cover many of the core aspects

of fiscal space: government debt sustainability, perceived sovereign risk, market access, balance sheet composition, external and private debt considerations. Among other indicators, Kose et al. (2017) reports also the so-called *de facto* fiscal space as derived in Aizenman and Jinjarak (2010) and Aizenman et al. (2013), which delineates the government's ability to raise tax revenues to contain public debt and deficits. We relate to Kose et al. (2017) and Aizenman and Jinjarak (2010) for the construction of our model-free proxies for fiscal room in the US. Moreover, in Auerbach (1997), but also in Gale and Auerbach (2009) and Auerbach and Gorodnichenko (2017), the authors approximate fiscal room in the US by measuring the size of fiscal distress via the government intertemporal budget constraint.⁵ We draw also from the approach described in Auerbach (1997) for building a measure of US fiscal space.

Another part of the literature focuses on providing structural macroeconomic approaches to assessing fiscal space. This literature builds theoretical models to derive the so-called *fiscal limit*. This line of work considers the fiscal space as the distance between current public debt and a (theory-based) debt or *fiscal* limit. The latter represents the maximum expected assets attainable by the government. On one hand, Ostry et al. (2010), Ghosh et al. (2013a) and Ostry et al. (2015) compute static estimates for debt limits based on the observation that the higher the levels of debt, the weaker the reaction of primary surpluses ("fiscal fatigue").^{6,7} On the other hand, in Bi (2012), Leeper (2013), Bi and Leeper (2013), Bi and Traum (2012) and Bi and Traum (2014), the theoretical fiscal limit corresponds to the discounted present value of future maximum primary surpluses.⁸ Moreover, Collard et al. (2015) also exploit the idea of a maximum primary surplus to derive a static measure of debt limit. We relate to this literature by exploiting the concept of maximum primary surplus and using it to build a time-varying measure for the fiscal room in the US.

Our paper is also related to the literature studying how the fiscal position affects the transmission of fiscal policy. A few papers have analyzed this aspect. On the theoretical side, in the seminal contribution from Perotti (1999), the author builds a simple model where government expenditure shocks have a positive, "non-Keynesian" correlation in fiscal stress times.

⁵More details are provided in Section 3.3.

⁶Their approach provides static debt limit and fiscal space estimates. These studies consider two calibration periods: from 1970 to 2007 and from 1985 to 2007. Moreover, updated fiscal space static estimates are regularly reported in Moody's (2011).

⁷Note that these debt limits cannot be retrieved by a model-free estimation.

⁸Maximum surpluses arise if the government can steer tax revenues at the peak points of the Laffer curve (Trabandt and Uhlig, 2011).

Symmetrically, tax shocks have a negative, Keynesian correlation in normal times and a positive, non-Keynesian correlation in fiscal stress times. Also two other existing models (Blanchard, 1990; Sutherland, 1997) formalize the non-Keynesian effects of tax hikes at high levels of public debt, but the model presented in Perotti (1999) allows for both tax and government spending shocks to have non-Keynesian effects on private consumption via the expectations channel. This study finds a strong evidence that expenditure shocks have Keynesian effects when the level of public debt or deficits is low, and non-Keynesian effects in the opposite circumstances. In short, the more burdensome the state of fiscal stress the more a positive spending shock will lead to a steeper path of future expected tax changes and, thus, to a lower present value of consumers' wealth. Moreover, Aloui and Eyquem (2019) develop a New Keynesian model with capital, distortionary taxes and public debt in which the zero lower bound (ZLB) constraint on the nominal interest rate may be binding. In the absence of ZLB, high steady-state levels of government debt to GDP lead to reduced output multipliers. Empirically, Huidrom et al. (2019) show that the government spending multiplier is low (and even negative) under weak fiscal position. Using a sample going from 1980 until 2014, Huidrom et al. (2019) estimate an Interacted Panel VAR including 33 countries and find a multiplier that is between 0 and 0.5 when the economy is in a weak fiscal position (high government debt). The authors show that the weak fiscal position affects the size of the fiscal multiplier via two main channels: the Ricardian (households' expectations) channel and the interest rate channel. In a similar fashion, Ilzetzki et al. (2013) study fiscal multipliers according to various dimensions in a large panel of countries spanning from the '60s until before the Great Financial crisis. Finally, Auerbach and Gorodnichenko (2017) employ a local projection model for 25 OECD countries to study the multiplier in different government debt states. Using a sample spanning from 2003 until 2017, they estimate significant government spending multipliers above (below) 0 in low (high) public debt states. All of the aforementioned papers aim at studying the interaction between fiscal sustainability and fiscal policy transmission. However, they all use the simple and narrower measure of debt-to-GDP ratio as a state variable, which cannot convey enough information on the available fiscal room. To the best of our knowledge, our paper is the first to adopt a broader concept of fiscal space to study the transmission of fiscal shocks.

More in general, our paper relates to the literature studying state-dependency in fiscal policy both empirically and theoretically. Such literature, however, has focused mainly on investigating the role of the business cycle as a state.

In macroeconomic theory, very few papers (e.g., Michaillat, 2014; Albertini et al., 2019) concentrated on how recessions and expansions affect the size of the fiscal multipliers. The study of business cycle-dependent fiscal multipliers parallels Keynesian theory: a government expenditure shock has a stronger expansionary outcome during a recession. Crowding out of private spending and investment is attenuated by the slack state. This fueled an extensive use of discretionary fiscal policy to counteract business cycle contractions.

The two main empirical contributions on this topic are, on the one hand Auerbach and Gorodnichenko (2012) and Auerbach and Gorodnichenko (2013) and on the other hand Ramey and Zubairy (2018). Such papers deliver very different conclusions and do not reach a consensus in empirical research regarding the effects of expansionary fiscal shocks under slacks and boom. Indeed, using a regime-switching VAR approach, Auerbach and Gorodnichenko (2012) and Auerbach and Gorodnichenko (2013) find large differences between multipliers in recessions and expansions.⁹ However, the adopted econometric model requires to assume for how many quarters the impulse response function should remain in each state of the economy. This could lead to distorted results in favor of an artificially higher multiplier in recession. By contrast, Ramey and Zubairy (2018) conclude that there is no difference in the size of multipliers across the business cycle. Using the Jordà (2005) local projection approach and high unemployment rate as proxy for recession, they show that government spending multipliers range between 0.3 and 0.8 no matter the state of the business cycle.¹⁰ The authors also find mixed evidence on the size of the fiscal multiplier at the zero lower bound.^{11,12}

⁹Indeed, the authors observe that the multiplier is much higher in recessions rather than in expansions. They report the multiplier to be as high as 2.5.

¹⁰However, depending on the shock series used to identify government spending innovations, the results vary. Employing Ramey (2011b,a) defense news shock series, the authors find no evidence for any difference between multipliers; while, using the Blanchard and Perotti (2002) shock series, they observe different multipliers across states. This is due to extremely low values of the multiplier in expansion rather than peaks in recessions.

¹¹In Ramey and Zubairy (2018), ZLB state is defined as the quarters in which the T-bill rate is equal to or below 50 basis points. When the authors use the full sample spanning from 1889:Q1 until 2015:Q4, the multiplier is not higher at the zero lower bound; while, excluding the World War II, they found a multiplier as high as 1.5 in the ZLB state.

¹²In Macroeconomic theory, Christiano et al. (2011); Coenen et al. (2012); Cogan et al. (2010) employ New-Keynesian DSGE models to show that fiscal multipliers tend to be higher when policy rates are at the ZLB.

Our paper draws from the approach developed in Ramey and Zubairy (2018) to estimate fiscal multipliers, but we instead find a major role for state dependency, suggesting that fiscal space could be the relevant state for the transmission of fiscal shocks.

3. FISCAL SPACE APPROXIMATIONS

As pointed out in Section 1, there exists no unique definition for fiscal space; its core aspect is to measure the debt service capacity of a country. The latter hinges on many dimensions: budget position, financing needs, spending and revenue prospects, resilience to contingent liabilities and access to markets. As correctly reported in Botev et al. (2016), fiscal space can be measured either in terms of achieving long-term sustainability or losing market access. In reality, risk premia arising from long-term sustainability concerns affect directly market access. Nonetheless, a single approach cannot comprehend all the aspects affecting fiscal space. Thus, the literature focuses either on long-term sustainability (Blanchard, 1990; Auerbach, 1997) or market access (Ghosh et al., 2013a). Botev et al. (2016) summarizes these two main dimensions of fiscal space altogether with the factors affecting it. According to one dimension, fiscal space is deemed as the distance between the actual debt and the debt limit for which the government would be unable to roll-over debt and, thus, lose market access. The other dimension characterizes fiscal space in terms of long-term s ustainability. Thus, for instance, a country with worrying health spending forecasts could have no fiscal space according to the long-term sustainability perspective, while having plenty of fiscal room in terms of market access. The lack of credit events in advanced economies complicates the estimation of fiscal space in terms of the market access dimension; the latter can take form only under structural macroeconomic approaches that build on theoretical assumptions to gauge debt limits.¹³ Therefore, our work will focus more on the long-term sustainability side of fiscal space, while still trying to embed aspects of the debt limit literature in the definition of our fiscal room measures.

In this section we describe our four historical indicators of fiscal space for the US. Two of them are based on model-free estimates of fiscal space, drawing from Kose et al. (2017), Aizenman and Jinjarak (2010) and Aizenman et al. (2013). Such measures focus on fiscal

¹³As far as we know, the only paper that manages to exploit the time-variation of sovereign credit data for advanced economies in order to estimate both debt limits and fiscal space in terms of the market access dimension is Pallara and Renne (2019).

sustainability, the revenue capacity of the government, fiscal policy stance and, indirectly, market access. One additional fiscal space proxy derives from the concept of fiscal imbalance as in Auerbach (1997). One final measure draws from the concept of fiscal (or debt) limit described in Bi (2012), Bi and Leeper (2013) and Ostry et al. (2010).

All of our indicators, as we clarify in the following sections, depend, among other key factors, i.e. on public debt, the surplus/deficit and, more in general, on government finance variables, all series that are highly correlated with expansions and recessions. Therefore, we adjust these series to purify the government's fiscal position from business cycle fluctuations.¹⁴ This correction ensures that we capture the discretionary dimension of fiscal room in the US together with its link to medium/long run economic *phenomena*.

3.1. **Primary surplus sustainability gap.** Our first indicator draws from Kose et al. (2017), which, in turn, builds on the debt accumulation accounting equation:

$$\Delta \frac{B_t}{Y_t} \approx \frac{i_t - \gamma_t}{1 + \gamma_t} \frac{B_{t-1}}{Y_{t-1}} - s_t,\tag{1}$$

where $\frac{B_t}{Y_t}$ is the debt-to-GDP, γ_t is the growth rate of nominal GDP, i_t is the nominal interest rate and s_t is the primary balance over GDP.¹⁵ Using this simple equation, we calculate the level of the primary surplus required, in each quarter, to stabilize public debt, i.e. to make $\Delta \frac{B_t}{Y_t} = 0$. We then define our fiscal space indicator as the distance between such primary surplus and the realized one. Using equation 1 and cyclically adjusting the variables as described in the previous section, our proxy is given by the following equation:

$$FS_{1,t} = s_t^{c.a.} - \left(\frac{i_t - \tilde{\gamma}_t}{1 + \tilde{\gamma}_t}\right) d_t^{c.a},\tag{2}$$

¹⁴Following the method implemented by the World Bank and reported in Kose et al. (2017), we cyclically adjust the government finance statistics variables by multiplying them by $(1 + \tilde{y})^{-(\epsilon_x - 1)}$ where \tilde{y} is the difference between the actual GDP and the CBO potential output as % of potential output; ϵ_x stands for the output gap elasticity of x for x = [Revenues, Government spending, Federal debt]. We use World Bank (see also Kose et al., 2017) elasticities for revenues and government spending, respectively equal to 1 and 0.1. For what concerns federal debt, we estimate the elasticity to be not significantly different from 0 and, thus, we assume it to be equal to 0. Note that the elasticities for spending and revenues proposed here are not distant from the estimates in Girouard and André (2006).

 $^{^{15}}$ Eq. 1 should include also the stock-flow adjustments not to hold as an approximation. Stock-flow adjustments comprise of factors that affect debt but are not included in the budget balance (such as acquisitions or sales of financial assets). For sake of simplicity, we focus on the "snowball-effect" side of the debt accumulation equation and on the government budget balance for the construction of FS_1 .

where $s_t^{c.a.}$ is the cyclically adjusted primary surplus over potential GDP, $d_t^{c.a}$ is the cyclically adjusted debt over potential GDP, $\tilde{\gamma}_t$ is the nominal potential GDP growth and i_t is the historical interest rate on 10-year maturity US government bonds.^{16,17} This gap incorporates information on the fiscal stance, public debt acceleration and the difference between the interest rate and GDP growth. This last measure represents the first nod to the study of fiscal sustainability in macroeconomics and contributes in a relevant way to the dynamics of our fiscal space indicator.¹⁸ We regard this measure as the benchmark indicator of fiscal space since it summarizes the many features a proxy for fiscal space should contain: considerations of fiscal sustainability, debt dynamics, interest rate, output growth and fiscal policy stance. In particular, this fiscal space measure highlights times of rapid debt accumulation due to inherent inability to roll-over debt via primary surpluses, crucial characteristics of the fiscal position of the government. We estimate FS_1 from 1889Q1 to explore the history of fiscal room in the United States. Figure 1 reports the results. The figure shows that fiscal space was especially tight over the depression of the early 20s, the Second World War and started to worsen from 2001 onward and, in particular, during the Great Financial Crisis.

3.2. Laffer curve peak-implied surplus gap. According to the literature on fiscal limits, fiscal space is defined by the distance between the actual debt and the maximum amount of debt the government can sustain, i.e. the debt limit. In Bi (2012) and Bi and Leeper (2013), this limit is defined by the discounted projected path of maximum primary surpluses implicit in the

¹⁶As potential GDP we take the CBO potential GDP estimates. Additionally, we also implemented sensitivity analysis using different potential GDP measures (e.g., sixth-degree polynomial for the logarithm of GDP) that leaves unaffected both the dynamics of the fiscal space measure and the econometric results of the paper.

¹⁷We acknowledge that the average debt maturity for the US is around 6 years and using the 5-year maturity yield would be more precise. However, we have only historical data points for the 10-year maturity yield, which correlates more than 95% with the 5-year maturity yield.

¹⁸According to Blanchard (2019), as long as the yields are lower than the GDP growth rate, even in the current low-growth environment, countries have fiscal space (see also Mauro and Zhou, 2019). However, there is a growing consensus that such argument is incomplete. For instance, the simple measure of i - g does not consider the evolution of the primary balance itself and the stock-flow adjustments. Moreover Jiang et al. (2019) find that the discount factor on government debt is decoupled from the yields on bonds, which would nuance the claims in Blanchard (2019).





Fiscal Space I (C.a. primary surplus sustainability gap)

The Figure shows the estimates for FS_1 or cyclically adjusted primary surplus sustainability gap in blue and its 1-year moving average in red. The series are expressed in annual percentage of potential output. The series are adjusted so that the shaded regions indicate the areas where fiscal space is tight (1-year FS > median).

peak of the Laffer curve.^{19,20} We exploit the concept of debt limit as described in Bi (2012) and Collard et al. (2015), but we apply this intuition on a quarter-by-quarter perspective. Indeed, we define and calculate below, for each quarter, the maximum primary surplus attainable in the US. We then define fiscal space as the difference between such maximum surplus and the

$$\ell_t = \mathbb{E}_t \left(\sum_{j=1}^{+\infty} \exp(\gamma_t - i_t + \dots + \gamma_{t+j-1} - i_{t+j-1}) s_{t+j}^* \right),$$
(3)

where ℓ_t is the fiscal limit-to-GDP, \mathbf{g}_t stands for nominal growth, i_t is the risk-free rate and s_t^* is the maximum surplus over GDP. The maximum primary surplus s_t^* is the surplus implicit at the peak of the Laffer curve.

¹⁹In Bi (2012) and Bi and Leeper (2013), fiscal limit is defined as follows,

²⁰The Laffer curve represents the reverse bell-shaped relationship between the average tax rate and government revenues.

realized primary deficit.²¹ This measure captures how far is the government from revenue maximization.

We calculate maximum government revenues using Laffer curve-peak tax rates estimates as in Trabandt and Uhlig (2011) for labour, capital and consumption:

$$\hat{T}_t = \tau_K^* T B_t^K + \tau_L^* T B_t^L + \tau_c^* T B_t^c,$$
(4)

where τ_{K}^{\star} , τ_{L}^{\star} and τ_{c}^{\star} stand for the peak tax rates.²² \hat{T}_{t} represents the maximum revenues and TB_{t}^{i} (i = K, L, c) are the tax bases. Then, we compute the (cyclically adjusted) maximum surplus as follows:²³

$$\hat{s}_t = (\hat{t}_t^{c.a.} - g_t^{c.a.}), \tag{5}$$

where $\hat{t}^{c.a.}$ and $g_t^{c.a.}$ are the cyclically adjusted maximum revenues and government spending, respectively. Finally, our fiscal space indicator is defined as:

$$FS_{2,t} = \hat{s}_t - deficit_t^{c.a.},\tag{6}$$

thus, the higher the distance, the greater the actual deficit compared to peak surplus \hat{s}_t .²⁴ Figure 2 reports the estimates for *FS*₂ expressed in percentage of potential output.

3.3. **Fiscal imbalance.** Auerbach (1997) first proposed a measure to quantify US fiscal imbalances by taking into account the impact of future government expenditure (such as contingent liabilities/implicit spending) and revenues on the intertemporal government budget constraint. The same approach has been employed in Gale and Auerbach (2009) and Auerbach and Gorodnichenko (2017). We adopt the idea of fiscal imbalance as described in Auerbach (1997) and we calculate, in each period, by what constant fraction of GDP taxes (revenues) would have to be increased for the government budget constraint to be satisfied when

²³See section 3 for details on the cyclical adjustment.

²¹Indeed, we can think of the debt-to-GDP ratio, $\frac{B_t}{Y_t}$, as the cumulative discounted stream of past deficits. Therefore, the distance between L_t and $\frac{B_t}{Y_t}$ gives the size of the available fiscal space of the country.

 $^{^{22}\}tau_{K}^{\star}$, τ_{L}^{\star} and τ_{c}^{\star} are respectively equal to 0.6, 0.52 and 0.05. These represent the tax rates estimated in the benchmark model in Trabandt and Uhlig (2011) for the US. τ_{L}^{\star} and τ_{k}^{\star} correspond to the benchmark Laffer curve model with Frisch elasticity equal to 3 and intertemporal elasticity of substitution equal to 2. For more details on the tax rates we refer to appendix A.2. Moreover, We decided to use ones of the lowest estimates because there is no explicit mention of compliance neither in the present work nor in Trabandt and Uhlig (2011), namely the higher the tax rate the more tax evasion and avoidance become relevant phenomena. Compliance is actually a meaningful issue concerning tax rates and revenues (Pappada and Zylberberg, 2017).

²⁴For instance, during WWII and the Great Recession, the Laffer curve peak-implied surplus was smaller than the actual deficit.

FIGURE 2. *FS*₂ (1929:2-2015:4): Laffer curve peak-implied surplus gap



Fiscal Space II (Laffer curve peak-implied surplus gap)

The Figure shows the plot for FS_2 in blue and its 1-year moving average in red. The series are demeaned and expressed in quarterly percentage of potential output. The series are adjusted so that the shaded regions indicate the areas where fiscal space is tight (1-year FS > median).

the dynamics of future spending is considered. In our approach, we estimate the fiscal imbalance measuring the fiscal adjustment needed to satisfy the government budget constraint over a 10 years horizon:

$$B_t = (1+i_t)^{-[(t+H)-t]} \left(\frac{B_t}{GDP_t}\right) GDP_{t+H} - \sum_{k=0}^{t+H} (1+i_t)^{-(k+1-t)} (S_k + \Delta_t GDP_k),$$
(7)

where B_t is the total nominal government debt, S_t is surplus, i_t is the interest rate on the ten years maturity government bond and H is the last horizon (10 years, namely 40 quarters). Δ_t represents the quarterly fiscal imbalance as a percentage of GDP. The government budget constraint implies a projected path for purchases, revenues and income. These projections account for the foreseen dynamics in implicit spending for healthcare and the social security system. For the out-of-sample forecasts we use CBO projections that consider spending for



Fiscal Space III (Fiscal imbalance)

The Figure shows the estimates for the fiscal imbalance measure \dot{a} la Auerbach. The series is expressed in quarterly percentage of potential output. The shaded regions indicate the areas where fiscal space is tight (*FS* > median).

health and pensions under current and anticipated regulations.²⁵ The advantage of this fiscal space measure lies in its forward-looking nature, as it considers the projected path of the government budget. Figure 3 reports the estimates for FS_3 expressed in percentage of potential output. The chart highlights the periods of major fiscal distress for the US government.

3.4. *De facto* fiscal space. Aizenman and Jinjarak (2010) and Aizenman et al. (2013) build a measure to capture fiscal room defined as *de facto* fiscal space. The measure proposed is inversely related to the tax-years necessary to repay public debt or deficits. *De facto* fiscal space in Aizenman and Jinjarak (2010), Aizenman et al. (2013), Kose et al. (2017) is defined as the ratio of either public debt or the deficit over GDP (averaged over several years) to

²⁵Since CBO projections are only available from 2006, for the in-sample projections, we take the realized values of the considered variables assuming perfect foresight.

the average *de facto* tax base, namely the realized tax collection averaged over several years in order to smooth for business cycle fluctuations. We build on the concept of *de facto* fiscal space to construct an indicator that uses current cyclically adjusted revenues and deficits run by the government. In our approach, we define *de facto* fiscal space in the following way:

$$FS_{4,t} = \frac{(deficit_t^{c.a.})}{(receipts_t^{c.a.})},$$
(8)

where $deficit_t^{c.a.}$ stands for the cyclically adjusted deficit-to-potential GDP and $receipts_t^{c.a.}$ represents the total realized government tax receipts over potential GDP.²⁶ The advantage of this measure is to provide insights on the actual tax capacity of a country to balance current deficits. *FS*₄ highlights periods of high deficit overhangs with respect to the government inability to raise revenues via tax collection. Figure 4 reports the estimates for *FS*₄.

3.5. Properties of the fiscal space indicators. For each fiscal space indicator, we generate a dummy variable equal to 1 (0) when our FS measures is above (below) its median value, meaning that fiscal space is tight (loose). In Table 1 we calculate the correlations among such dummies, in order to show how our indicators relate to each other. Dummy indicators FS_1 and FS_4 show the highest correlation (equal to 0.73 in median). FS_2 and FS_4 also display significant correlation (equal to 0.64 in median) since government revenues are key in their respective fiscal space definitions. Dummies FS_2 and FS_3 show very low median correlation (equal to 0.01) and we cannot reject that they are uncorrelated. The remaining crosscorrelations among fiscal space dummies are above 30% in median. These results suggest that our indicators are mutually consistent while capturing different aspects of the evolution of fiscal space. Table 2 reports the correlations between fiscal space dummies and a broad set of relevant macroeconomic indicators, in particular NBER recession dates, zero lower bound (ZLB) dates and high/low federal debt-to-GDP ratio periods. The correlation between all four fiscal space measures and the NBER recession dates is approximately null, consistent with the fact that our proxies are purified from the transitory effects of the business cycle.²⁷ Our fiscal space measures are rather linked to the discretionary dimension of government

²⁶We use latest CBO potential output estimates and we also estimated the fiscal space using a sixth-degree polynomial for the logarithm of GDP as real trend GDP as a robustness, which did not lead to changes in the dynamics of the estimates neither the interpretation of the fiscal space size over the sample.

 $^{^{27}}$ This is evident also from Figure 13 in Appendix E, which reports tight fiscal space periods across our *FS* dummies and recession events over time.

FIGURE 4. FS_4 (1889:1-2015:4): De Facto fiscal space



Fiscal Space IV (De Facto Fiscal Space)

The Figure shows the estimates for our definition of the *de facto* fiscal space *á la* Aizenman and Jinjarak (2010). The plot for FS_4 is in blue and its 1-year moving average in red. The series expresses the relative size of deficit with respect to government revenues. Shaded areas stand for tight fiscal space periods (1-year FS > median).

finance variables and to medium/long run economic phenomena.²⁸ Tight fiscal space states also partially relate to ZLB periods.²⁹ This is mainly due to the Second World War and the Great Recession, in which both fiscal distress and low interest rates coexisted. However, although such correlations are positive, they are not particularly high. Table 2 also shows that our measures are related to high/low government debt-to-GDP periods.³⁰ Such correlations range between 0.24 and 0.7, highlighting the role of public debt in the evolution of fiscal

²⁸In Table 36 in the Appendix, we also report the correlations of fiscal space series with unemployment and potential output.

²⁹ZLB dummy indicates the state when the interest rate is at the zero lower bound or the FED is being very accommodative of fiscal policy (1932Q1-1951Q1, 2008Q4-2015Q4).

³⁰High (low) federal debt-to-GDP ratio means that the federal debt-to-GDP ratio is above (below) its historical median, which is equal to 32.85%.

space.³¹ However, debt-to-GDP is not sufficiently informative to capture the whole dynamics of fiscal space: other key factors such as output trends, the interest rate and the fiscal policy stance are also of crucial importance for identifying the level of fiscal room in the economy. Finally, in Table 2 we also calculate the correlations of our indicators with a dummy variable equal to one when the US was involved in a major war and with another dummy variable capturing the party of the US president. None of these correlations are relevant, suggesting that our fiscal space indicators are not driven neither by military spending nor by the political cycle.

In Figure 5, we plot the periods in which fiscal space is identified as tight, according to each *FS* dummy. The panel at the center of the figure reports a similar information, however using debt-to-GDP, which is the standard indicator used by the literature to identify periods of fiscal distress. Two main results emerge from Figure 5. First, our measures are well related with each other, as most of the periods identified are common across the four indicators. This is especially the case for FS_1 and FS_4 . Second, tight fiscal space periods identified by debt-to-GDP do not coincide with those identified by our method. Indeed, the concept of fiscal space refers to a broader notion of fiscal sustainability as opposed to the debt-to-GDP ratio. In fact, fiscal space takes into account the dynamics of other key macroeconomic variables and the fundamental debt capacity of the economy.

	Median	Lower bound	Upper bound
$\mathbb{C}orr(FS1_t^d, FS2_t^d)$	0.597	0.520	0.673
$Corr(FS1_t^d, FS3_t^d)$	0.344	0.248	0.439
$Corr(FS1_t^d, FS4_t^d)$	0.732	0.670	0.798
$Corr(FS2_t^d, FS3_t^d)$	0.014	-0.090	0.121
$Corr(FS2_t^d, FS4_t^d)$	0.636	0.564	0.707
$Corr(FS3_t^d, FS4_t^d)$	0.326	0.234	0.424

TABLE 1. Bootstrapped correlations (95 % confidence interval) among Fiscal space dummies.

The Table shows the estimates for median non-parametric bootstrapped correlation coefficients and their intervals at the 95% confidence level. Intervals are calculated using the normal approximation. FSj_t^d (j = 1, 2, 3, 4) stand for the fiscal space dummies.

³¹Additionally, in Table 36 in the Appendix, we can see that the fiscal space series are consistently correlated with debt-to-GDP series and the change in public debt.



FIGURE 5. FS dummies and High Debt-to-GDP ratio.

This figure shows the periods of Tight Fiscal Space via our *FS* state dummies and of high Debt-to-GDP ratio (identifying high debt periods whenever Debt-to-GDP ratio is above its median). See also Table 2 for more details.

	Median	Lower bound	Upper bound
$\mathbb{C}orr(FS1_t^d, Rec_t^d)$	-0.165	-0.247	-0.083
$\mathbb{C}orr(FS2^d_t, Rec^d_t)$	-0.074	-0.112	0.096
$\mathbb{C}orr(FS3^d_t, Rec^d_t)$	0.170	0.071	0.273
$\mathbb{C}orr(FS4^d_t, Rec^d_t)$	-0.070	-0.158	0.024
$\mathbb{C}orr(FS1_t^d, \frac{B}{GDPt})$	0.487	0.407	0.565
$\mathbb{C}orr(FS2_t^d, \frac{B}{GDPt})$	0.545	0.425	0.665
$Corr(FS3_t^d, \frac{B}{GDP_t}^d)$	0.243	0.108	0.371
$\mathbb{C}orr(FS4^d_t, \frac{B}{GDP_t}^d)$	0.701	0.603	0.810
$\mathbb{C}orr(FS1_t^d, ZLB_t^d)$	0.349	0.277	0.420
$\mathbb{C}orr(FS2^d_t, ZLB^d_t)$	0.047	-0.053	0.150
$\mathbb{C}orr(FS3^d_t, ZLB^d_t)$	0.394	0.302	0.484
$\mathbb{C}orr(FS4^d_t, ZLB^d_t)$	0.341	0.268	0.417
$\mathbb{C}orr(FS1_t^d, War_t^d)$	0.089	0.004	0.172
$\mathbb{C}orr(FS2^d_t, War^d_t)$	-0.047	-0.158	0.060
$\mathbb{C}orr(FS3^d_t, War^d_t)$	0.130	0.026	0.227
$\mathbb{C}orr(FS4^d_t, War^d_t)$	0.110	0.026	0.197
$\mathbb{C}orr(FS1_t^d, Dem_t^d)$	0.169	0.083	0.255
$\mathbb{C}orr(FS2^d_t, Dem^d_t)$	-0.027	-0.133	0.076
$\mathbb{C}orr(FS3^d_t, Dem^d_t)$	0.081	-0.025	0.185
$\mathbb{C}orr(FS4^d_t, Dem^d_t)$	0.146	0.061	0.227

TABLE 2. Bootstrapped correlations (95 % confidence interval) between Fiscal space dummies and NBER recession dates, High/Low Debt-to-GDP states, ZLB periods, War dates and US political cycle.

The Table shows the estimates for median non-parametric bootstrapped correlation coefficients and their intervals at the 95% confidence level. Intervals are calculated using the normal approximation. FSj_t^d (j = 1, 2, 3, 4) are the fiscal space dummies, Rec_t^d is the dummy for NBER recession dates $\frac{B}{GDP_t}^d$ is the dummy for high-low federal debt-to-GDP states^{*a*}, ZLB_t^d is the dummy for the zero lower bound state, War_t^d is the dummy indicating US involvement in major wars.^{*b*} Lastly, Dem_t^d stands for the dummy indicating the party of the US president in the office each quarter.

^{*a*}High (low) federal debt status is considered as above (below) the historical median.

Finally, to further validate our fiscal space indicators, we also provide a narrative behind their evolution over time. Both dummies FS_1 and FS_4 indicate that the fiscal space was tight between 1917 and the end of 1920 (see Figures 1 and 4). This is due to the large wartime increase in the debt-to-GDP ratio, which rose from a 3% level up to 30% on average. Moreover, this tight fiscal space state partially overlaps with the depression of 1920-21 characterized by

^bWe consider as major wars involving the US the following conflicts: Spanish-American War (1898), Philippine-American War (1899-1902), World War I (1914-1918), World War II (1939-1945), Korean War (1950-1953), Vietnam War (1965-1973), Gulf War (1990-1991), Afghanistan War (started in 2001), Iraq War (2003-2011), American-led intervention in Syria and Iraq (2014-present).

extreme deflation and whose key factor was the erroneous tightening stance by the FED.³² Not surprisingly, all indicators identify the Great Depression and the Second World War as periods of tight fiscal space. Indeed, these periods are characterized by sluggish growth, high real interest rates, increasing debt-to-GDP ratios and deficits, all ingredients generating a reduction in fiscal space. FS_1 , FS_2 and FS_4 dummies correctly signal the military build-ups due to the Korean war in 1953-54 (see Figures 1, 2 and 4). Additionally, virtually all fiscal space dummies (Figures 1, 2, 3 and 4) identify a long-lasting tight fiscal space state starting with the 1973 Oil crisis and continuing through the 1979 energy crisis, the fiscal expansionary policies during the Reagan administration and the Gulf war. Lastly, as already mentioned, the *FS* dummies characterize the Great Financial Crisis and its aftermath as tight fiscal space periods, given the hefty rise in public spending, the low-growth and low-inflation environment.³³

Notably, in Figure 5, we observe that indicators FS_1 and FS_4 report periods of tight fiscal space between the end of 1916 and the beginning of 1920, while debt-to-GDP was low (and below its historical median). Indeed, in this time-span, deficit over GDP rose from being close to inexistent to oscillating around 15% until 1919. Our indicators righteously pick up the unsustainable fiscal path of government finances of those years compared to the erroneous signal of the debt stock. Similarly, from the end of the 70*s* until the beginning of the 80*s*, our indicators signal a period of tight fiscal space while debt-to-GDP was at low levels (around 30%). During this period, the distance between the cost of financing debt (namely, the yield on government bonds) and the growth rate of output (see discussion in Sec. 3.1) oscillated around 5% (capping in 1982 at 10%).³⁴ This points out both that the economy was slowing down, due to the energy crisis, and that United States experienced its most volatile money growth rates in the post-war era, which translated into higher yields. Additionally, in the late 90*s* and early 2000*s*, while debt-to-GDP was higher than 60% (and than its historical median), our indicators denote periods of loose fiscal space. This is due to the evolution of deficit over

³²For details, see Friedman and Schwartz (2008)

³³Public spending rose following the fiscal stimulus packages enacted from 2008 onwards. First, the Economic Stimulus Act of 2008 (enacted February 13, 2008) was an Act of Congress providing for several kinds of economic stimuli intended to boost the United States economy in 2008 and to avert a recession, or ameliorate economic conditions. Second, the American Recovery and Reinvestment Act of 2009 was a stimulus package signed into law by President Barack Obama in February 2009. The approximate cost of this stimulus package was estimated to be \$ 831 billion.

³⁴As in Section 3.1 we use the 10-year maturity yield.

GDP, which decreased from an average 5% in the early 90s to being mildly negative (surplus) at the end of the decade before rising again from 2002 onwards to early 90s levels.

4. METHODOLOGY

In this section we present the methodology employed to estimate the fiscal multipliers depending on the state of fiscal s pace. We then describe the identification method adopted and we outline the specification of the empirical model.

4.1. **State-dependent Local Projection.** Local Projections - introduced in Jordà (2005) - are becoming an increasingly popular estimation strategy for Impulse Response Functions (IRF) as opposed to more standard methods like structural VARs. A wide range of estimation procedures can in principle be applied to estimate LPs, and our approach hinges on a standard IV strategy to identify the relevant IRFs. Nevertheless, the discussion to follow is general enough to be applied to other estimation procedures. In a general form, the kind of linear Local Projections we are interested in estimating can be written as

$$y_{t+h} = \alpha_h + \beta_h g_t + \psi_h(L) \mathbf{X}_{t-1} + \varepsilon_{t+h} \quad h = 0, 1, 2, \dots, H,$$
(9)

where y_t is the variable whose dynamic response we want to track, g_t is the endogenous variable we want to shock (government spending in our application), and X_t is a vector of control variables. Estimation is performed *separately* for each horizon and for each dependent variable with two-stage least squares. Generally speaking, IRFs are defined by the sequence $\{\beta_h\}_{h=0}^H$, and inference is performed with Newey-West standard errors.

The focus of this paper is on state-dependent responses of macroeconomic variables to fiscal p olicy. The non-linearity we add is a very simple one, i.e. we investigate the extent to which fiscal policy is transmitted differently under two different regimes, and we separate those two states with a simple indicator variable. Specifically,

$$y_{t+h} = S_{t-1} [\alpha_{A,h} + \beta_{A,h}g_t + \psi_{A,h}(L)\mathbf{X}_{t-1}] + (1 - S_{t-1}) [\alpha_{B,h} + \beta_{B,s}g_t + \psi_{B,h}(L)\mathbf{X}_{t-1}] + \varepsilon_{t+h}$$
(10)
$$h = 0, 1, 2, \dots, H.$$

This kind of non-linearity is conceptually the same as the one used in (e.g.) Ramey and Zubairy (2018).³⁵ Other authors - e.g. Tenreyro and Thwaites (2016) - have opted for smooth transition local projections, which allow parameters to smoothly switch between the two regimes, instead of letting them change abruptly around a threshold. While a smooth transition is desirable, for this model - first developed in Granger et al. (1993) - to be employed one needs to calibrate two key curvature and location parameters, whose choice turns out to be quite important in terms of the final set of IRFs that are obtained. In principle, those parameters could be estimated, but in order to do so reliably the researcher would need a lot of data around the transition of the state variable, something that is virtually never the case in macroeconomic applications.³⁶ We therefore decided to stick with the easier to interpret (and more robust) discrete indicator variable, which nonetheless yields a cleaner interpretation of the coefficients as exact average causal effects within a given state.

4.2. Model specification and Identification. In our approach, we estimate the LP model as in equation 10. The state dependency is given by the lagged dummy variable S_{t-1} that indicates the fiscal space status. Taking as baseline proxy FS_1 (Eq. 2 in section 3.1), we define the fiscal space state as tight (large) whenever the 1 year moving average proxy is above (below) its historical median.^{37,38} We implement an IV approach using two different shock series to instrument government spending g_t in equation 10. The first shock series that we use as IV is Ramey news. Ramey (2011b,a) builds a series of estimated changes in expected present value of government purchases caused by military events: the so-called Ramey news shock series.³⁹ The second series that we consider as instrument for government spending is the Blanchard and Perotti shock. Blanchard and Perotti (2002) provide identification for both government

 $^{^{35}}$ The underlying assumption in this framework is that, once we calculate impulse response function in each state, the IRFs remain in that same state through the whole horizon of the estimation. Such assumption seems plausible in our analysis given the persistent behaviour of our fiscal space indicators. Indeed, the average duration of periods characterized by tight fiscal space is, respectively for FS_1 , FS_2 , FS_3 , FS_4 , of 31.5, 17.6, 43.5, 23.3 quarters, suggesting the slow moving behaviour of fiscal room. In section 5 we also calculate the impulse response of the state itself, showing how the dynamics of the states over time are not particularly affected by the shock in the short term.

³⁶Teräsvirta (1994) discusses those estimation issues in detail.

³⁷We take the 1 year averaged series so to have a smooth enough series and make sure that the fiscal space state is persistent and lasts at least one year. The average quarters spent in tight fiscal space are approximately equal to 20.

³⁸The same definition for fiscal space state goes for FS_2 and FS_4 indicators proposed in Section 3; while, given the intrinsic smoothness of the series, for FS_3 (see section 3) there is no need to take the 1-year average. Then, the derived fiscal space state series are used to support the results obtained in section 5.

³⁹In these papers, the author embeds the derived series in a VAR ordered first (Choleski recursive causal ordering).

spending and tax shocks in a structural VAR, where government spending is ordered first. The identification is based on short-run restrictions and on the automatic stabilizers of fiscal policy to economic activity. A similar approach has been employed also by Fatás et al. (2001) and Galí et al. (2007) among others.

We follow the Ramey and Zubairy (2018) approach to compute fiscal multipliers.⁴⁰ The authors propose to scale output, government spending and the shock series by trend GDP in order to estimate directly the multipliers.^{41,42} Additionally, as proposed in Mountford and Uhlig (2009), we estimate integral multipliers.⁴³ In order to estimate the cumulative output, consumption and investment multipliers, we adopt a one-step IV estimation of

$$\sum_{j=0}^{h} y_{t+j} = S_{t-1}[\alpha_{A,h} + m_{A,h} \sum_{j=0}^{h} g_{t+j} + \psi_{A,h}(L)\mathbf{X}_{t-1}] + (1 - S_{t-1})[\alpha_{B,h} + m_{B,h} \sum_{j=0}^{h} g_{t+j} + \psi_{B,h}(L)\mathbf{X}_{t-1}] + u_{t+h},$$
(11)

instrumenting government spending with the Ramey news and Blanchard and Perotti shock series. Under this approach, we can estimate the integral state-dependent multiplier $m_{i,h}$ (i = A, B) in one step. This allows us to calculate directly the standard errors of the multipliers and, therefore, to implement statistical inference. We select a lag-order of 4 as in most of the empirical fiscal policy literature (e.g., Fatás et al., 2001; Blanchard and Perotti, 2002; Galí et al., 2007; Ramey and Zubairy, 2018, among others) that accounts for the implementation lags of fiscal policy. An advantage of adopting LP methods compared to VARs is the possibility to include a wider set of controls without worrying about the quadratic increase of parameters in the estimation. Hence, we include an extensive set of controls (in X_{t-1} in equations 10 and 11) that include the average marginal tax rate as in Barro and Redlick (2011) and Bernardini and Peersman (2018), the nominal interest rate on 10 years maturity government bonds, the logarithm of the implicit GDP deflator, real consumption and real investment scaled by trend GDP, the ratio of federal debt to lagged GDP, the ratio of current government deficits on GDP

⁴⁰One of the pitfalls of calculating multipliers: the use of log transformed variables (as in most fiscal VAR analyses) requires an *ex-post* adjustment that leads to biased multipliers. The multipliers must be multiplied by the average output-to-spending ratio that greatly varies for the US across different estimation samples. As observed in Ramey and Zubairy (2018), In post-WWII, the average output-to-spending ratio is equal to 5; while, from 1890 the average output-to-spending ratio is equal to 8.

⁴¹And, by analogy, any real variables whose multiplier is of interest for the researcher.

⁴²The real GDP time trend is estimated as a sixth-degree polynomial for the logarithm of GDP.

⁴³Rather than peak multipliers as in Blanchard and Perotti (2002) or average multipliers given the initial shock as in Auerbach and Gorodnichenko (2012).

and, lastly, the corporate bond spread (AAA Moody's - Y10). In the bag of controls, lags of output, government spending and the shock series are included. Details on the data used can be found in Appendix A.⁴⁴

5. Results

This section presents the results, reporting the state-dependent effects of government spend-ing shocks. First we provide the main result of the paper, obtained using FS_1 as the baseline state variable for measuring fiscal space. However, later we provide additional evidence us-ing the remaining fiscal space proxies to prove the stability of our results, together with a further robustness section regarding the sample size.

5.1. **IRF and fiscal multipliers.** Figure 6 reports the impulse response function for government spending and economic activity when fiscal shock is estimated using the Ramey news, both in the linear case (top panel) and in the two states of large and tight fiscal space (bottom panel). First of all, we note that the evolution of the two variables of interest in the linear case is fairly standard and in line with that reported in Ramey and Zubairy (2018). Consistent with the dynamics of a news shock, actual government consumption slowly increases and peaks around 10 quarters after the initial impulse. Economic activity follows a comparable pattern. Turning to the non-linear case, we observe a similar dynamic in the two states, even if the same reaction in GDP in the large fiscal space case is determined by a less pronounced increase in spending. However, in order to evaluate quantitatively the effects of fiscal policy, both in general and in particular in non-linear cases, graphical impulse response functions are not particularly useful, as we need to compare shocks of similar size and take into account the evolution of the instrumented variable (government consumption in this case). In line with the literature (Ramey and Zubairy, 2018; Mountford and Uhlig, 2009) we calculate the cumulative fiscal multipliers and we henceforth concentrate on this measure to quantify the impact of spending shocks. Tables 3 and 4 report the fiscal multipliers at each horizon, respectively for the Ramey shock and the Blanchard-Perotti shock. The first column of the tables presents the value of the fiscal multiplier in the linear case, while the second and the third column in large and tight fiscal s pace. Finally, the last column tests the statistical significance of the difference between the multipliers in the two states, reporting the p-value of the test. Table 3 shows that, while the linear case presents multipliers smaller than one,

⁴⁴All scaled by trend GDP.

this average effect is very different once disentangled between our two states. Indeed, in tight fiscal space the multiplier averages around 0.6, while in large fiscal space is around 1.5. Such difference is present at each horizon and it is always statistically significant.⁴⁵ A similar picture emerges from Table 4, which finds slightly larger fiscal multipliers in the two states. However, the main takeaway remains valid and represents the principal result of the paper, being the fiscal multiplier smaller than one in the tight fiscal space state and larger than one in the opposite case. Such result is particularly important as, adopting a methodology which follows closely that of Ramey and Zubairy (2018) we draw very different conclusions regarding the state dependent nature of fiscal policy. We also note that such results are not driven by an unbalanced distribution of shocks in the two states. Indeed, as we show in Appendix C, shocks are equally distributed between periods of tight fiscal space and periods of large fiscal space.⁴⁶ Tables 5 and 6 calculate fiscal multipliers when we change the way we define our two states. Section 3.5 clarified that, for each of our measures, we define fiscal space as tight (large) when the indicator is above (below) the median. Tables 5 and 6 report the results when instead we concentrate on extreme episodes, meaning that we define fiscal space as tight when the underlying indicator (FS_1 in this case) is above the 80th percentile and as large when it is below the 20th. We do so to investigate whether our results depend upon the threshold adopted to distinguish between the two states. Both tables show results consistent with the main takeaway of the paper. Indeed, the effects are even more pronounced, in line with what one would expect looking at the extreme tails of the distribution. In particular, while the fiscal multiplier becomes larger as there is more fiscal room available, the multiplier shrinks when we consider periods of very tight fiscal space.

⁴⁵However, as the relevance of Ramey instrument is lower for the first horizons of the impulse response, we concentrate our attention on the period two-years after the shock

⁴⁶We cannot instead exclude the possibility of composition effects occurring if shocks to government consumption and investment are distributed unevenly in periods of tight and large fiscal space. Unfortunately, there is not a proper way to check for this possibility. Ramey news have been constructed through a careful work consisting of analyzing weekly newspapers and magazines in the search of military spending news, without distinguishing whether the future increase/decrease in military spending regards consumption or investments. Moreover, the standard practice when adopting this identification method is to instrument general government spending, which comprises both government consumption and investment. While we acknowledge that this strategy could confound two possibly different types of shocks, such problem is common to every paper adopting this identification scheme. Moreover, as government investment shocks are much less frequent than consumption shocks, we believe that the possible confounding effect should not alter the estimates by a too large factor.



The Figure shows the estimates for the impulse response function of real government spending scaled by trend output (*G*) and real GDP scaled by trend output (*GDP*) following a Ramey news shock.

5.2. **Transmission mechanism.** In this subsection we investigate the transmission mechanism. Figure 15 reports the impulse response of private consumption and investment. While in the linear case they both fall in response to a positive government spending shock, when state-dependencies are considered we observe an opposite behavior in the two states. When fiscal space is tight, private consumption and investment decreases; by contrast, when fiscal space is large they both increase, giving rise to non-Ricardian effects. This result is made clear in Table 7, which reports the consumption and investment multipliers, together with the associated error bands. The consumption multiplier in the large fiscal space state (red line) is around 0.5 one-year after the shock and slowly decays over time; in the tight fiscal space it is negative, around -0.1, and slowly reverts towards zero. A similar pattern, although less pronounced, holds for private investment. When fiscal space is large the multiplier is basically zero, given the wide uncertainty surrounding the estimates; when fiscal space is tight,

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63154	2.59324	0.20505	0.0013***
2	(0.14822)	(0.82358)	(0.24826)	
2-years	0.73058	1.84459	0.48422	0.0009***
-	(0.15028)	(0.33708)	(0.17141)	
3-years	0.84010	1.50515	0.60074	$5.88 \cdot 10^{-8^{***}}$
-	(0.11044)	(0.11772)	(0.13488)	
4-years	0.86660	1.32009	0.61816	$9.18 \cdot 10^{-6^{***}}$
2	(0.10107)	(0.12729)	(0.14161)	
5-years	0.96984	1.33614	0.60957	0.0002***
-	(0.17138)	(0.13926)	(0.18746)	
Signif. codes	0.01***	0.05**	0.1^{*}	

TABLE 3. Fiscal Space-dependent Fiscal Multiplier: Baseline - FS_1 - Ramey News Shock (1929-2015)

TABLE 4. Fiscal Space-dependent Fiscal Multiplier: Baseline - FS_1 - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.58461	1.17810	0.59973	0.14563
2	(0.07333)	(0.34638)	(0.07025)	
2-years	0.72999	1.19385	0.81420	0.27954
	(0.10334)	(0.31641)	(0.12329)	
3-years	0.77180	1.49097	0.85638	0.00059***
	(0.11598)	(0.13626)	(0.14276)	
4-years	0.82710	1.51589	0.91605	0.00017^{***}
	(0.14259)	(0.14345)	(0.17982)	
5-years	0.79798	1.71866	0.90564	0.00868***
	(0.18411)	(0.34744)	(0.19070)	
Signif. codes	0.01***	0.05**	0.1^{*}	

instead, private investment multiplier is negative and significant. The differential effects highlighted in Figure 15 and Table 7 are at the root of the difference in the output multipliers reported in the previous section. The mechanism we have in mind to rationalize this empirical evidence relates to Perotti (1999), which shows how the degree of public finance sustainability alters the transmission of fiscal policy.⁴⁷ Indeed, a deficit-financed increase in government spending generates an increase in future taxation, needed to repay the cost of

⁴⁷In his seminal contribution Perotti (1999) refers to public debt as the variable defining the state of public finance, as opposed to fiscal space. Moreover, Perotti (1999) studies an economy populated both by unconstrained and constrained individuals. However, the main intuition of the paper applies irrespective of such choices.





The Figure shows the estimates for the integral multiplier of real consumption and real total private investment.

the fiscal expansion. However, in an environment with distortionary taxation, such tightening will be more pronounced when fiscal space is already tight, because of the convexity in tax distortions. This steeper path of future taxes, internalized by agents, produces a larger negative wealth effect and therefore a subdued reaction of private consumption. Although we believe the aforementioned channel is at the heart of the differential evolution of private consumption in the two states, it is not straightforward to obtain supporting empirical evidence, given the forward-looking nature of the variables involved, in particular expectations regarding future taxation.⁴⁸

For the sake of completeness, we also analyze the response of all control variables employed in regression 10. Figures 18 and 19 report the results. The average marginal tax rate (AMTR),

⁴⁸Another potential way to interpret our results is that, in an economy with finite-lived agests, in the tight fiscal space state we obtain standard Ricardian effects on consumption. At the opposite, in the large fiscal space state the negative wealth effect following the fiscal shock is small as consumers perceive they will not have to repay all of the government spending during their finite lifetime, therefore pushing consumption up.

at least for a large part of the horizon of the IRF, does not display significant different behaviour in the two states. Although such result would suggest that tax do not play a major role in explaining the dichotomy in fiscal multipliers, we stress that the horizon of our IRF is at most five years, too short to observe changes in future tax rates, also given the length of the political cycle⁴⁹. The figure also shows the evolution of debt and deficit ratios, which both increase much more in the tight fiscal space. Such behavior suggests that economic growth and the evolution of budget variables, in the case of large fiscal space, contribute positively to repay for the initial fiscal stimulus and avoid to generating a large increase in the long-run overall level of debt-to-GDP. On the contrary, in the tight fiscal space state, debt and deficit are magnified by the less favourable environment following the shock. This is clear also from Figure 17, where we report the estimates for the debt multiplier; the latter proves to be significantly higher in the tight state with respect to large fiscal space over the impulse response horizon.⁵⁰ Turning to other variables of interest for the transmission of fiscal shock, the most important one is the interest rate. Consistently with the puzzling behavior of the linear case, already shown in the empirical literature by Mountford and Uhlig (2009), Fisher and Peters (2010) and Ramey (2011b), the nominal interest rate slightly decreases in both states.^{51,52} Finally, the response of the price deflator falls slightly in the tight fiscal space state, while it remains constant in the other state.⁵³

5.3. Additional results and robustness. This section proposes a series of additional results and robustness checks. First of all, we show that all of our results are robust to different definitions of fiscal space. In order to do so, we perform the same estimations as those provided in the previous section, however adopting the remaining measures, FS_2 , FS_3 and FS_4 , to identify periods of tight and large fiscal space. In all these cases, fiscal multipliers are calculated

⁴⁹Results from Perotti (1999) link explicitly the effect of government spending to the political cycle: a lower probability of survival of the policy maker implies a steeper path of future expected taxation, which is distortionary, and, thus, a larger negative wealth effect following a public expenditure shock.

⁵⁰In Appendix **B**, we show the adjustment needed for the correct computation of the debt multiplier.

⁵¹Mountford and Uhlig (2009) observe that a government spending shock reduces investment, although interestingly not via higher interest rates (that are moderately falling).

⁵²Referring to Fisher and Peters (2010) and Ramey (2011b), Murphy and Walsh (2016) report that one possible explanation for the fall in interest rates is an endogenous response of monetary policy to government spending shocks.

⁵³Figure 20 in Appendix E reports the IRFs of the state variables itself, i.e. when $FS_{1,t}$ is considered as dependent variable. The figure shows that the responses are not very significant in large fiscal space, while fiscal space narrows even more in the tight state. This suggests that our states do not change regime over the impulse response horizon, validating the econometric framework adopted to investigate the state-dependency (as described in Section 4.1).
using both Ramey shocks and Blanchard-Perotti ones. Tables 7 and 8 summarize, respectively for the two identification methods, the fiscal multiplier when fiscal space is given by measure FS_2 . The multipliers in the two states are statistically significant, with those in the large fiscal space state being consistently above one and those in the alternative state below one. A similar narrative emerges from Tables 9 and 10, which adopt FS_3 as fiscal space indicator. According to these estimates, the fiscal multiplier in the tight fiscal space is basically zero at each horizon given the large error bands associated, with a point-estimate which in some cases turns even negative. The multiplier in large fiscal space is instead statistically significant and greater than one. Tables 11 and 12 repeat the same exercise using FS_4 and once again confirm the main result of the paper. All in all, these results confirm the importance of considering fiscal space for the transmission of fiscal policy. Moreover, they re-assure that the methods employed to measure the evolution of fiscal space over time are consistent, as different indicators in the end produce very similar results. Finally, in this robustness section we study how a different sample size affects our results. We first show the results when the estimation is performed only in the post-WWII period and after we reconsider the full sample (1929-2015) once we exclude the global financial crisis of the late 2000s. Tables 13, 14, 15 and 16 reports the fiscal multiplier, respectively for FS_1 , FS_2 , FS_3 and FS_4 , when the estimation is performed over the period 1947-2015. The government spending shock is estimated only using Blanchard-Perotti shocks, as it is not possible to use the Ramey shock on such a short sample. As Ramey and Zubairy (2018) show, such shocks do not have enough variation to be relevant when instrumenting government spending in the more recent period. The aforementioned tables provide the same univocal picture. Indeed, when considering only the post-war period the difference among multipliers in the two states is magnified, with multipliers in the tight fiscal space shrinking towards zero and instead becoming larger than 1.5 when public finances are sound. Finally, Tables from 17 to 24 calculate fiscal multipliers over the full sample for each fiscal space indicator and each identification method, omitting the great financial crisis. In order to do so, we exclude from the sample the period 2007:Q4 -2010:Q4 Results clearly show that the financial crisis does not play a role in the determining the size of fiscal multipliers, as results remain basically unchanged with respect to the baseline. In Appendix D, we report additional estimates of fiscal multipliers to further validate our findings.⁵⁴

⁵⁴In Appendix D, we report estimates of the fiscal multiplier across states of recession and expansion (Tables 34 and 35) finding no evidence for a difference in multipliers as in Ramey and Zubairy (2018). Moreover,

6. CONCLUSIONS

The paper investigated the state-dependent effects of fiscal policy, once the dynamics of fiscal sustainability and fiscal room are jointly considered. Drawing from different strands of the literature we developed several indicators of fiscal space and we measured its evolution over time. The main result highlighted by this paper is that fiscal space matters for the transmission of fiscal policy, as fiscal multipliers are much larger (smaller) when fiscal shocks are implemented in periods of loose (tight) fiscal space. Such a result appears important mainly in two respects. First of all, it stresses the importance of state-dependency in the study of fiscal policy. While the recent literature has found only minor differences in fiscal multipliers across business cycle and monetary policy regimes, our paper finds that, by contrast, fiscal space matters a lot. Second, the paper shows, especially from a policy prospective, that fiscal policy can be a very powerful tool in stimulating the economy, but this is not always the case. Particular attention needs to be paid to the economic conditions in which fiscal policy is implemented, as weak public finances could hamper the transmission of fiscal shocks and, in extreme cases, even produce detrimental effects.

we also compute the multiplier across states of High/Low Federal Debt-to-GDP (Tables 28 and 29) and velocity (Tables 30 and 31) that validate our results even though the estimates are mostly not significant. Lastly, we also show the results on the fiscal multiplier interacting our baseline tight fiscal space state with the zero lower bound without getting any relevant result (see Tables 32 and 33).





The Figure shows the estimates for the impulse response function of real consumption scaled by trend output (C) and real total private investment scaled by trend output (I) following a Ramey news shock.





Mult. P. Debt (Large/Tight FS)



The Figure shows the estimates for the integral multiplier of federal debt. The correct multiplier is derived by adjusting the resulting 2SLS, see Appendix B for details.





The Figure shows the estimates for the impulse response function of the average marginal tax rate as in Barro and Redlick (2011) and Bernardini and Peersman (2018) (*AMTR*), federal debt-to-lagged GDP, deficit-to-GDP following a Ramey news shock.



FIGURE 11. log(IPGDP), Y10, C.B. spread IRFs - FS_1 - Ramey News (1929-2015)

The Figure shows the estimates for the impulse response function of the logarithm of the implicit price deflator (log(IPGDP)), 10-years government bond yield (Y10), corporate bond spread (C.B. spread) following a Ramey news shock.

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63154	2.81727	-0.21469	$3.50 \cdot 10^{-7^{***}}$
2	(0.14822)	(0.56600)	(0.13276)	
2-years	0.73058	1.70203	0.35317	0.00120***
-	(0.15028)	(0.40783)	(0.07934)	
3-years	0.84010	1.88123	0.58868	$6.01 \cdot 10^{-7^{***}}$
2	(0.11044)	(0.25094)	(0.06006)	
4-years	0.86660	1.77096	0.54652	$1.56 \cdot 10^{-9^{***}}$
5	(0.10107)	(0.19341)	(0.07147)	
5-years	0.96984	1.80390	0.42486	$3.53 \cdot 10^{-16^{***}}$
5	(0.17138)	(0.14593)	(0.08855)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 5. Fiscal Space-dependent Fiscal Multiplier: Extremes - FS_1 - Ramey News Shock (1929-2015)

TABLE 6. Fiscal Space-dependent Fiscal Multiplier: Extremes - FS_1 - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.58461	2.14032	0.35654	$4.71 \cdot 10^{-6^{***}}$
2	(0.07333)	(0.37591)	(0.09213)	
2-years	0.72999	2.23721	0.55709	$3.40 \cdot 10^{-12^{***}}$
2	(0.10334)	(0.22677)	(0.08843)	
3-years	0.77180	2.37341	0.60591	0***
-	(0.11598)	(0.10159)	(0.07333)	
4-years	0.82710	2.29881	0.53609	0***
	(0.14259)	(0.07158)	(0.08137)	
5-years	0.79798	3.49345	0.31733	0.00225***
	(0.18411)	(1.03655)	(0.07357)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.64555	1.88273	0.44398	0.06724*
5	(0.15678)	(0.77452)	(0.10089)	
2-years	0.74378	1.74077	0.76256	0.03579**
2	(0.15380)	(0.45041)	(0.06955)	
3-years	0.84977	1.76922	0.85130	0.03082**
-	(0.11385)	(0.44278)	(0.04943)	
	(0.10841)	(0.38137)	(0.04137)	
4-years	0.87766	1.59097	0.85816	0.02053**
-	(0.11009)	(0.31850)	(0.03960)	
5-years	0.98316	1.47619	0.92338	0.01052**
ž	(0.14459)	(0.20549)	(0.04105)	
Signif. codes	0.01***	0.05**	0.1^{*}	

TABLE 7. Fiscal Space-dependent Fiscal Multiplier: FS_2 - Ramey News Shock (1929-2015)

TABLE 8. Fiscal Space-dependent Fiscal Multiplier: FS_2 - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.57740	0.78042	0.59651	0.58361
5	(0.07776)	(0.32745)	(0.05936)	
2-years	0.72720	1.41969	0.83229	0.04505*
-	(0.10753)	(0.29852)	(0.06061)	
3-years	0.77148	2.01395	0.88710	0.04079*
	(0.12386)	(0.48205)	(0.05796)	
4-years	0.82850	2.23201	0.90586	0.11401
	(0.15460)	(0.68862)	(0.05945)	
5-years	0.80963	2.43213	0.84064	0.14908
	(0.19480)	(1.06717)	(0.05386)	
Signif. codes	0.01***	0.05**	0.1^{*}	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63154	0.06175	-1.29681	0.40762
2	(0.14822)	(0.59070)	(1.45076)	
2-years	0.73058	1.05849	-0.24363	0.00357***
-	(0.15028)	(0.22630)	(0.35900)	
3-years	0.84010	1.55694	0.23091	0.00010***
	(0.11044)	(0.19291)	(0.28947)	
4-years	0.86660	1.59637	0.11623	0.06148^{*}
-	(0.10107)	(0.28190)	(0.73807)	
5-years	0.96984	1.41966	5.82865	0.94841
-	(0.17138)	(0.39664)	(68.12317)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 9. Fiscal Space-dependent Fiscal Multiplier: FS_3 - Ramey News Shock (1929-2015)

TABLE 10. Fiscal Space-dependent Fiscal Multiplier: FS_3 - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-years	0.58461	0.76187	0.45005	0.19568
2	(0.07333)	(0.25133)	(0.08753)	
2-years	0.72999	1.05299	0.55512	0.13296
-	(0.10334)	(0.25256)	(0.24944)	
3-years	0.77180	1.43210	0.62393	0.15728
	(0.11598)	(0.24250)	(0.54125)	
4-years	0.82710	1.48427	1.22196	0.81256
	(0.14259)	(0.27920)	(1.06721)	
5-years	0.79798	1.59052	0.92785	0.17297
	(0.18411)	(0.34296)	(0.35007)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63154	3.75108	0.28423	0.01406**
2	(0.14822)	(1.38169)	(0.22158)	
2-years	0.73058	2.05270	0.60125	0.04503**
-	(0.15028)	(0.69020)	(0.13488)	
3-years	0.84010	1.88193	0.73149	0.05170*
	(0.11044)	(0.56384)	(0.11566)	
4-years	0.86660	1.85698	0.75159	0.04717**
	(0.10107)	(0.49944)	(0.11650)	
5-years	0.96984	1.95285	0.79650	0.02287**
	(0.17138)	(0.41647)	(0.12164)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 11. Fiscal Space-dependent Fiscal Multiplier: FS_4 - Ramey News Shock (1929-2015)

TABLE 12. Fiscal Space-dependent Fiscal Multiplier: FS_4 - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.58461	1.10744	0.62750	0.22530
2	(0.07333)	(0.41681)	(0.07530)	
2-years	0.72999	1.81899	0.79186	0.01629**
-	(0.10334)	(0.45954)	(0.10872)	
3-years	0.77180	2.49842	0.80025	0.00201***
-	(0.11598)	(0.56230)	(0.13046)	
4-years	0.82710	2.68677	0.85427	0.00054^{***}
	(0.14259)	(0.52123)	(0.17516)	
5-years	0.79798	2.64423	0.79593	0.00022***
	(0.18411)	(0.43731)	(0.20899)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.78241	1.58795	0.00138	0.00035***
2	(0.25513)	(0.28862)	(0.42515)	
2-years	0.84343	1.57688	-0.25010	0.00297***
-	(0.25494)	(0.23773)	(0.63100)	
3-years	0.98088	1.67524	-0.75365	0.00878***
	(0.24902)	(0.19679)	(0.90610)	
4-years	1.02607	1.93252	-0.70540	0.03675**
	(0.26606)	(0.20905)	(1.23985)	
5-years	1.08399	2.17235	-0.27979	0.04273**
	(0.25559)	(0.18201)	(1.18732)	
Signif. codes	0.01***	0.05**	0.1^{*}	

TABLE 13. Fiscal Space-dependent Fiscal Multiplier: Baseline - FS_1 - Blanchard and Perotti Shock (1947-2015)

TABLE 14. Fiscal Space-dependent Fiscal Multiplier: FS_2 - Blanchard and Perotti Shock (1947-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.78985	1.10333	0.79735	0.47290
2	(0.25330)	(0.29539)	(0.32461)	
2-years	0.84899	1.83026	0.89623	0.02108**
-	(0.25321)	(0.33419)	(0.27809)	
3-years	0.98172	1.84778	0.93012	0.01249**
-	(0.24859)	(0.31810)	(0.28208)	
4-years	1.02114	1.76111	0.89232	0.00982***
-	(0.26731)	(0.20896)	(0.28746)	
5-years	1.07650	1.82689	0.57381	0.00254***
	(0.25816)	(0.15332)	(0.38968)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.78241	1.28176	-1.13708	0.02744**
2	(0.25513)	(0.28592)	(1.04374)	
2-years	0.84343	1.41547	-2.04633	0.00022***
-	(0.25494)	(0.35766)	(0.86894)	
3-years	0.98088	1.60183	-3.90825	$3.12 \cdot 10^{-9^{***}}$
2	(0.24902)	(0.33893)	(0.89952)	
4-years	1.02607	1.62012	-6.53132	$1.19 \cdot 10^{-7^{***}}$
2	(0.26606)	(0.48563)	(1.53528)	
5-years	1.08399	1.57074	16.14953	0.07216*
-	(0.25559)	(0.52679)	(8.18320)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 15. Fiscal Space-dependent Fiscal Multiplier: FS_3 - Blanchard and Perotti Shock (1947-2015)

TABLE 16. Fiscal Space-dependent Fiscal Multiplier: FS_4 - Blanchard and Perotti Shock (1947-2015)

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.78241	1.56535	-0.00647	0.00265***
2	(0.25513)	(0.34450)	(0.40649)	
2-years	0.84343	1.76408	-0.26973	0.00292**
-	(0.25494)	(0.33348)	(0.57273)	
3-years	0.98088	1.95465	-0.78494	0.00074^{***}
	(0.24902)	(0.29949)	(0.73521)	
4-years	1.02607	1.95133	-0.90096	0.00536***
	(0.26606)	(0.29333)	(0.88961)	
5-years	1.08399	1.98488	-0.68068	0.02237**
	(0.25559)	(0.31918)	(0.96942)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63896	2.55736	0.06951	0.00074***
, ,	(0.14066)	(0.71521)	(0.27292)	
2-years	0.74396	1.73903	0.35627	0.00055***
-	(0.15074)	(0.34519)	(0.16722)	
3-years	0.85485	1.41452	0.47287	$1.54 \cdot 10^{-8^{***}}$
2	(0.09304)	(0.12955)	(0.14483)	
4-years	0.87870	1.27713	0.40662	0.00002***
-	(0.11012)	(0.12861)	(0.22632)	
5-years	0.97504	1.30148	0.30150	0.01104**
	(0.12487)	(0.14879)	(0.42682)	
Signif. codes	0.01***	0.05**	0.1^{*}	

TABLE 17. Fiscal Space-dependent Fiscal Multiplier: Baseline FS_1 Ramey News shock - Omit Crisis

TABLE 18. Fiscal Space-dependent Fiscal Multiplier: Baseline FS_1 Blanchard and Perotti shock - Omit Crisis

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.54651	1.14114	0.53379	0.16038
2	(0.07806)	(0.40017)	(0.08380)	
2-years	0.67261	1.21673	0.69745	0.22398
-	(0.11321)	(0.34065)	(0.13816)	
3-years	0.70251	1.62939	0.71931	0.00001***
-	(0.13626)	(0.17793)	(0.13992)	
4-years	0.75703	1.58922	0.79008	$1.16 \cdot 10^{-6^{***}}$
2	(0.16345)	(0.17858)	(0.18488)	
5-years	0.71765	1.80372	0.77752	0.00621***
2	(0.24502)	(0.45168)	(0.22075)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63896	1.94393	0.44715	0.01460**
2	(0.14066)	(0.58502)	(0.13637)	
2-years	0.74396	1.86138	0.71612	0.00533***
-	(0.15074)	(0.40031)	(0.06716)	
3-years	0.85485	1.80166	0.80316	0.05382*
-	(0.09304)	(0.49519)	(0.05158)	
4-years	0.87870	1.66456	0.85050	0.01837**
-	(0.11012)	(0.32840)	(0.04357)	
5-years	0.97504	1.54585	0.92032	0.02200**
-	(0.12487)	(0.26892)	(0.05141)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 19. Fiscal Space-dependent Fiscal Multiplier: FS_2 Ramey News shock - Omit Crisis

TABLE 20. Fiscal Space-dependent Fiscal Multiplier: FS_2 Blanchard and Perotti shock - Omit Crisis

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.54651	0.76758	0.49637	0.42038
2	(0.07806)	(0.32950)	(0.05274)	
2-years	0.67261	1.26224	0.71170	0.07070^{*}
-	(0.11321)	(0.31148)	(0.05126)	
3-years	0.70251	1.64182	0.78839	0.00305***
-	(0.13626)	(0.28023)	(0.04261)	
4-years	0.75703	1.99602	0.80259	0.00787^{*}
-	(0.16345)	(0.44734)	(0.04848)	
5-years	0.71765	3.54341	0.74643	0.53239
-	(0.24502)	(4.48613)	(0.06969)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63896	0.27217	-1.89126	0.42092
5	(0.14066)	(0.59703)	(2.53122)	
2-years	0.74396	1.16356	-0.32423	0.00670***
-	(0.15074)	(0.23563)	(0.46379)	
3-years	0.85485	1.60680	0.37605	0.00003***
	(0.09304)	(0.20433)	(0.22082)	
4-years	0.87870	1.65308	0.33197	0.00806***
-	(0.11012)	(0.30131)	(0.39214)	
5-years	0.97504	1.48477	-1.14925	0.63974
	(0.12487)	(0.41332)	(5.60615)	
Signif. codes	0.01***	0.05**	0.1^{*}	

TABLE 21. Fiscal Space-dependent Fiscal Multiplier: FS_3 Ramey News shock - Omit Crisis

TABLE 22. Fiscal Space-dependent Fiscal Multiplier: FS_3 Blanchard and Perotti shock - Omit Crisis

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
5	0.54651	0.79592	0.39108	0.09210*
	(0.07806)	(0.24045)	(0.11518)	
9	0.67261	1.10331	0.43783	0.03523**
	(0.11321)	(0.24385)	(0.22482)	
13	0.70251	1.46096	0.43690	0.00878^{***}
	(0.13626)	(0.23693)	(0.29317)	
17	0.75703	1.50260	1.07389	0.17138
	(0.16345)	(0.27861)	(0.17707)	
21	0.71765	1.59887	1.13084	0.27203
	(0.24502)	(0.33966)	(0.28495)	
Signif. codes	0.01***	0.05**	0.1*	

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.63896	4.85276	0.13557	0.06720*
5	(0.14066)	(2.58132)	(0.22769)	
2-years	0.74396	2.38234	0.46661	0.00518***
-	(0.15074)	(0.72015)	(0.12042)	
3-years	0.85485	2.06588	0.62613	0.01986**
-	(0.09304)	(0.61290)	(0.14324)	
4-years	0.87870	1.96149	0.60623	0.01136**
-	(0.11012)	(0.51298)	(0.15202)	
5-years	0.97504	2.07480	0.58778	0.00236***
	(0.12487)	(0.45158)	(0.22400)	
Signif. codes	0.01***	0.05**	0.1^{*}	

TABLE 23. Fiscal Space-dependent Fiscal Multiplier: FS_4 Ramey News shock - Omit Crisis

TABLE 24. Fiscal Space-dependent Fiscal Multiplier: FS_4 Blanchard and Perotti shock - Omit Crisis

Horizon	Linear	Large FS	Tight FS	<i>p</i> -value Diff.
1-year	0.54651	1.14853	0.55512	0.15520
-	(0.07806)	(0.40505)	(0.08777)	
2-years	0.67261	1.79567	0.69595	0.02435**
-	(0.11321)	(0.48749)	(0.12827)	
3-years	0.70251	2.37696	0.69215	0.00233***
-	(0.13626)	(0.54977)	(0.16690)	
4-years	0.75703	2.86686	0.75256	0.00029***
-	(0.16345)	(0.58986)	(0.19760)	
5-years	0.71765	2.81454	0.71643	0.00011***
-	(0.24502)	(0.53381)	(0.22694)	
Signif. codes	0.01***	0.05**	0.1*	

References

- Aizenman, J., Hutchison, M., and Jinjarak, Y. (2013). What is the risk of european sovereign debt defaults? fiscal space, cds spreads and market pricing of r isk. *Journal of International Money and Finance*, 34:37–59.
- Aizenman, J. and Jinjarak, Y. (2010). De facto fiscal space and fiscal stimulus: Definition and assessment. Working Paper 16539, National Bureau of Economic Research.
- Albertini, J., Auray, S., Bouakez, H., Eyquem, A., et al. (2019). Taking off into the wind: Unemployment risk and state-dependent government spending multipliers. Technical report.
- Aloui, R. and Eyquem, A. (2019). Spending multipliers with distortionary taxes: Does the level of public debt matter? *Journal of Macroeconomics*, 60:275–293.
- Auerbach, A. J. (1997). Quantifying the current us fiscal imbalance. Technical report, National Bureau of Economic Research.
- Auerbach, A. J. and Gorodnichenko, Y. (2012). Measuring the output responses to fiscal policy. *American Economic Journal: Economic Policy*, 4(2):1–27.
- Auerbach, A. J. and Gorodnichenko, Y. (2013). Output spillovers from fiscal policy. *American Economic Review*, 103(3):141–46.
- Auerbach, A. J. and Gorodnichenko, Y. (2017). Fiscal stimulus and fiscal sustainability. Technical report, National Bureau of Economic Research.
- Barro, R. J. and Redlick, C. J. (2011). Macroeconomic effects from government purchases and taxes. *The Quarterly Journal of Economics*, 126(1):51–102.
- Bernardini, M. and Peersman, G. (2018). Private debt overhang and the government spending multiplier: Evidence for the united states. *Journal of Applied Econometrics*, 33(4):485–508.
- Bi, H. (2012). Sovereign Default Risk Premia, Fiscal Limits, and Fiscal Policy. *European Economic Review*, 56(3):389–410.
- Bi, H. and Leeper, E. M. (2013). Analyzing Fiscal Sustainability. Staff working papers, Bank of Canada.
- Bi, H. and Traum, N. (2012). Estimating Sovereign Default Risk. *American Economic Review, Papers & Proceedings*, 102(3):161–166.
- Bi, H. and Traum, N. (2014). Estimating Fiscal Limits: The Case of Greece. *Journal of Applied Econometrics*, 29(7):1053–1072.
- Blanchard, O. (2019). Public debt and low interest rates. *American Economic Review*, 109(4):1197–1229.

- Blanchard, O. and Perotti, R. (2002). An Empirical Characterization of the Dynamic Effects of Changes in Government Spending and Taxes on Output. *The Quarterly Journal of Economics*, 117(4):1329–1368.
- Blanchard, O. J. (1990). [can severe fiscal contractions be expansionary? tales of two small european countries]: Comment. *NBER macroeconomics annual*, 5:111–116.
- Botev, J., Fournier, J.-M., and Mourougane, A. (2016). A re-assessment of fiscal space in oecd countries.
- Christiano, L., Eichenbaum, M., and Rebelo, S. (2011). When is the government spending multiplier large? *Journal of Political Economy*, 119(1):78–121.
- Coenen, G., Erceg, C. J., Freedman, C., Furceri, D., Kumhof, M., Lalonde, R., Laxton, D., Lindé, J., Mourougane, A., Muir, D., et al. (2012). Effects of fiscal stimulus in structural models. *American Economic Journal: Macroeconomics*, 4(1):22–68.
- Cogan, J. F., Cwik, T., Taylor, J. B., and Wieland, V. (2010). New keynesian versus old keynesian government spending multipliers. *Journal of Economic dynamics and control*, 34(3):281– 295.
- Collard, F., Habib, M., and Rochet, J.-C. (2015). Sovereign Debt Sustainability In Advanced Economies. *Journal of the European Economic Association*, 13(3):381–420.
- Escolano, M. J. (2010). A practical guide to public debt dynamics, fiscal sustainability, and cyclical adjustment of budgetary aggregates. International Monetary Fund.
- Fatás, A., Mihov, I., et al. (2001). *The effects of fiscal policy on consumption and employment: theory and evidence*, volume 2760. Citeseer.
- Fisher, J. D. and Peters, R. (2010). Using stock returns to identify government spending shocks. *The Economic Journal*, 120(544):414–436.
- Friedman, M. and Schwartz, A. J. (2008). *A monetary history of the United States*, 1867-1960. Princeton University Press.
- Gale, W. G. and Auerbach, A. J. (2009). The economic crisis and the fiscal crisis: 2009 and beyond. *Tax Notes*, 125(1).
- Galí, J., López-Salido, J. D., and Vallés, J. (2007). Understanding the effects of government spending on consumption. *Journal of the european economic association*, 5(1):227–270.
- Ghosh, A., Kim, J. I., Mendoza, E., Ostry, J., and Qureshi, M. (2013a). Fiscal Fatigue, Fiscal Space and Debt Sustainability in Advanced Economies. *Economic Journal*, pages F4–F30.

- Ghosh, A. R., Ostry, J. D., and Qureshi, M. S. (2013b). Fiscal Space and Sovereign Risk Pricing in a Currency Union. *Journal of International Money and Finance*, 34:131 163.
- Girouard, N. and André, C. (2006). Measuring cyclically-adjusted budget balances for oecd countries. *Available at SSRN 2005002*.
- Gordon, R. J. and Krenn, R. (2010). The end of the great depression 1939-41: Policy contributions and fiscal multipliers. Technical report, National Bureau of Economic Research.
- Granger, C. W., Terasvirta, T., et al. (1993). Modelling non-linear economic relationships. *OUP Catalogue*.
- Heller, M. P. S. (2005). Understanding fiscal space. International Monetary Fund.
- Huidrom, R., Kose, M. A., Lim, J. J., and Ohnsorge, F. L. (2019). Why do fiscal multipliers depend on fiscal positions? *Journal of Monetary Economics*.
- Ilzetzki, E., Mendoza, E. G., and Végh, C. A. (2013). How big (small?) are fiscal multipliers? *Journal of monetary economics*, 60(2):239–254.
- Jiang, Z., Lustig, H., Van Nieuwerburgh, N. S., and Xiaolan, M. Z. (2019). The us government risk premium puzzle.
- Jordà, Ó. (2005). Estimation and inference of impulse responses by local projections. *American economic review*, 95(1):161–182.
- Kose, A., Kurlat, S., Ohnsorge, F., and Sugawara, N. (2017). A Cross-Country Database of Fiscal Space. CEPR Discussion Papers 12196, C.E.P.R. Discussion Papers.
- Leeper, E. M. (2013). Fiscal Limits and Monetary Policy. NBER Working Papers 18877, National Bureau of Economic Research, Inc.
- Ley, E. (2009). Fiscal policy for growth.
- Mauro, P. and Zhou, J. (2019). r-g<0: Can we sleep more soundly? *IMF Work in progress*.
- Mertens, K. and Ravn, M. O. (2013). The dynamic effects of personal and corporate income tax changes in the united states. *American economic review*, 103(4):1212–47.
- Michaillat, P. (2014). A theory of countercyclical government multiplier. *American Economic Journal: Macroeconomics*, 6(1):190–217.
- Moody's (2011). Fiscal Space. Special Report Economic & Consumer Credit Analytics, Moody's Analytics.
- Mountford, A. and Uhlig, H. (2009). What are the effects of fiscal policy shocks? *Journal of applied econometrics*, 24(6):960–992.

- Murphy, D. and Walsh, K. J. (2016). Spending shocks and interest rates. Technical report, Mimeo.
- Ostry, J., Ghosh, A., Kim, J. I., and Qureshi, M. (2010). Fiscal Space. IMF Staff Position Notes 2010/11, International Monetary Fund.
- Ostry, J. D., Ghosh, A. R., and Espinoza, R. A. (2015). When Should Public Debt Be Reduced? IMF Staff Discussion Notes, International Monetary Fund.
- Pallara, K. and Renne, J.-P. (2019). Fiscal limits and sovereign credit spreads. *Available at SSRN* 3475425.
- Pappada, F. and Zylberberg, Y. (2017). Austerity and tax compliance. *European Economic Review*, 100:506–524.
- Perotti, R. (1999). Fiscal policy in good times and bad. *The Quarterly Journal of Economics*, 114(4):1399–1436.
- Perotti, R. (2007). *Fiscal policy in developing countries: a framework and some questions*. The World Bank.
- Ramey, V. A. (2011a). Can government purchases stimulate the economy? *Journal of Economic Literature*, 49(3):673–85.
- Ramey, V. A. (2011b). Identifying government spending shocks: It's all in the timing. *The Quarterly Journal of Economics*, 126(1):1–50.
- Ramey, V. A. and Zubairy, S. (2018). Government spending multipliers in good times and in bad: evidence from us historical data. *Journal of Political Economy*, 126(2):850–901.

Shiller, R. J. (1992). Market volatility. MIT press.

- Sutherland, A. (1997). Fiscal crises and aggregate demand: can high public debt reverse the effects of fiscal policy? *Journal of public economics*, 65(2):147–162.
- Tenreyro, S. and Thwaites, G. (2016). Pushing on a string: Us monetary policy is less powerful in recessions. *American Economic Journal: Macroeconomics*, 8(4):43–74.
- Teräsvirta, T. (1994). Specification, estimation, and evaluation of smooth transition autoregressive models. *Journal of the american Statistical association*, 89(425):208–218.
- Trabandt, M. and Uhlig, H. (2011). The Laffer Curve Revisited. *Journal of Monetary Economics*, 58(4):305–327.

APPENDIX A. DATA

A.1. **Data table.** In Table 25, we show the time series used both for the estimation of our fiscal space proxies in Section 3 and the empirical analysis described in Section 4. We provide also the sources where the data are retrieved with relative samples, the sections where the series are used and the transformation applied to them for both Sections 3 and 4.

TABLE 25. Data table

Variable	Sample	Source	Transformation & Usage
Government Spending (Real and Nominal)	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 3; Sec. 4: scaled by real trend GDP
Government Revenues (Real and Nominal)	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 3; Sec. 4: nominal over nom. GDP
Nominal GDP	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. <mark>3</mark> ; Sec. <mark>4</mark>
Real GDP	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 3; Sec. 4
Real Consumption	1919:Q1-1946:Q4	Gordon and Krenn (2010)	Sec. 4: growth rate
Real Consumption	1947:Q1-2015:Q4	$FRED^{a}$	Sec. 4
Real Investment	1919:Q1-1946:Q4	Gordon and Krenn (2010)	Sec. 3; Sec. 4: growth rate
Real Investment	1947:Q1-2015:Q4	FRED	Sec. <mark>3</mark> ; Sec. <u>4</u>
T-bill rate	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 3; Sec. 4
10-y Gov. bond yield	1889:Q1-1952:Q4	Shiller (1992)	Sec. <mark>3</mark> ; Sec. 4
10-y Gov. bond yield	1953:Q1-2015:Q4	Bloomberg	Sec. 3; Sec. 4
Federal debt	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. <mark>3</mark> ; Sec. 4 : nominal over lag nom. GDP
Deficit	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 3; Sec. 4: nominal over nom. GDP
Implicit GDP deflator	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 3; Sec. 4: in logarithm
Real Potential Output	1889:Q1-2015:Q4	CBO^b	Sec. 3
Nominal Potential Output	1889:Q1-2015:Q4	CBO	Sec. 3
Government spending forecasts	2006:Q1-2015:Q4	CBO	Sec. 3
Government revenues forecasts	2006:Q1-2015:Q4	CBO	Sec. 3
Federal debt forecasts	2006:Q1-2015:Q4	CBO	Sec. 3
Real trend GDP ^c	1889:Q1-2015:Q4	Ramey and Zubairy (2018)	Sec. 4
Average marginal tax rate	1919:Q1-1949:Q4	Barro and Redlick (2011)	Sec. 4: Federal individual income tax, stacking
Average marginal tax rate	1950:Q1-2013:Q4	Mertens and Ravn (2013)	Sec. 4: All tax units (series 1)
NBER recession dates	1919:Q1-2015:Q4	FRED	Sec. 3
Unemployment rate		Ramey and Zubairy (2018)	Sec. 4 ; Sec. 3
Dividends	1929:Q1-1946:Q4	HSUS^d	Sec. 3: cubic spline interpolation
Dividends	1947:Q1-2015:Q4	FRED	Sec. 3
Corporate profits before taxes	1929:Q1-1946:Q4	HSUS	Sec. 3: cubic spline interpolation
Corporate profits before taxes	1947:Q1-2015:Q4	FRED	Sec. 3
Gross wages and salaries	1929:Q1-1946:Q4	HSUS	Sec. 3: cubic spline interpolation
Gross wages and salaries	1947:Q1-2015:Q4	FRED	Sec. 3
Moody's Seasoned Aaa Corporate Bond Yield	1929:Q1-2015:Q4	FRED	Sec. 4
Moody's Seasoned Baa Corporate Bond Yield	1929:Q1-2015:Q4	FRED	Sec.4

^{*a*}Federal Reserve Economic Data (St. Louis FED)

^bCongressional Budget Office

^cThe real GDP time trend is estimated as a sixth-degree polynomial for the logarithm of GDP, from 1889Q1 through 2015Q4.

^{*d*}Historical Statistics of the United States. Series from HSUS are at annual frequencies and they are interpolated to quarterly frequencies by cubic spline.

A.2. FS_2 : Maximum tax rates and tax base series. In order to compute approximated government maximum revenues, we use the peak tax rates as derived in Trabandt and Uhlig (2011) to compute an approximation of the government maximum revenues. In their paper, the authors characterize the Laffer curves for capital and labour quantitatively for the US and several EU countries by comparing the balanced growth paths of a neoclassical growth model with constant Frisch elasticity preferences. Moreover, the authors implement a dynamic scoring analysis to explore how tax revenues and production adjust when labour and/or capital income taxes change and which portion of labour and/or capital tax cuts is self-financing. The Laffer curve for consumption taxes does not have a peak and is always increasing (approaching a tax rate of infinity). Hence, we replicate their results using an intertemporal elasticity of substitution (η in Table 26) equal to 2 and a Frisch elasticity (φ in Table 26) equal to 3. For what concerns the consumption tax rate, we take the nearest half-point maximum rate among the tax rates reported in Table 27 as it appears in Trabandt and Uhlig (2011). The maximum tax rates for labour (τ_n), capital (τ_k) are reported in Table 26. Lastly, we take dividends and corporate profits before taxes, wages and salaries before taxes and the portion of disposable income not destined to savings as proxies for the tax base series for capital, labour and consumption respectively. The data sources used for the tax base series are the Historical Database for the United States (HSUS) and FRED (see Table 25 in Appendix A for details).

TABLE 26. Characterization of US Laffer Curves for capital and labour (Dynamic Scoring at steady state, $\eta = 2$ and $\varphi = 3$).

% se	elf-fin.	ma	x. τ_k	max. add. tax rev.		
same	varied	same	varied	same	varied	
60	56	60	65	4	5	
% se	elf-fin.	ma	x. τ_n	max. add. tax rev.		
same	varied	same	varied	same	varied	
49	47	52	53	14	16	

TABLE 27. Consumption tax rates in % across years (Trabandt and Uhlig, 2011).

	1995	1996	1997	1998	1999	2000	2001	2002	2003	2004	2005	2006	2007
USA	5.1	5.1	5.0	5.0	4.9	4.8	4.6	4.5	4.5	4.4	4.5	4.4	4.2

APPENDIX B. DEBT MULTIPLIER ADJUSTMENT

Since we compute the multipliers for output, consumption, investment and public debt using the technique described in Section 4, the latter multiplier needs an *ex-post* adjustment given that public debt is not scaled by trend GDP (real). First, we assume that

$$\sum_{h=1}^{H} \frac{Debt_{t+h}^{N}}{GDP_{t-1+h}^{N}} \approx \sum_{h=1}^{H} \frac{Debt_{t+h}^{R}}{GDP_{t-1+h}^{R}},$$
 (Ass. I)

where the superscripts *N* and *R* indicate nominal and real variables, respectively. Second, we assume that the ratio of real GDP (but also its lag) and trend GDP is approximately constant,

$$\frac{GDP_{t-1+j}^{R}}{Trend_{t+j}^{GDPR}} \approx \kappa_{h}, \ j = 0, \dots, h$$
(Ass. II)

Indeed, κ_h oscillates around 1 over the impulse response horizon with very little variation. Thus, following the strategy described in Section 4, we regress cumulative debt over lagged GDP (nominal) on cumulative government spending scaled by trend output (real) as follows:

$$\sum_{j=0}^{h} \frac{Debt_{t+j}^{N}}{GDP_{t-1+j}^{N}} = \alpha_{h} + m_{h} \sum_{j=0}^{h} g_{t+j} + \psi_{h}(L) \mathbf{X}_{t-1} + u_{t+h},$$
(B.I)

where m_h represents the one-step cumulative multiplier for debt for each h. However, we are interested in finding the multiplier \hat{m}_h such that, for each h,

$$\sum_{j=0}^{h} \frac{Debt_{t+j}^{R}}{Trend_{t-1+j}^{GDP^{R}}} = \hat{\alpha}_{h} + \hat{m}_{h} \sum_{j=0}^{h} g_{t+j} + \hat{\psi}_{h}(L) \mathbf{X}_{t-1} + \hat{u}_{t+h}.$$
 (B.II)

We know that

$$\hat{m}_{h} = \frac{\partial \sum_{j=0}^{h} \frac{Debt_{t+j}^{R}}{Trend_{t+j}^{GDPR}}}{\partial \sum_{j=0}^{h} g_{t+j}}$$
(B.III)

and that

$$m_{h} = \frac{\partial \sum_{j=0}^{h} \frac{Debt_{t+j}^{N}}{GDP_{t-1+j}^{N}}}{\partial \sum_{j=0}^{h} g_{t+j}}$$
(B.IV)

thus, we can rewrite Eq. B.III as follows,

$$\hat{m}_{h} = \frac{\partial \sum_{j=0}^{h} \frac{Debt_{t+j}^{R}}{Trend_{t+j}^{GDPR}}}{\partial \sum_{j=0}^{h} g_{t+j}} = \frac{\partial \sum_{j=0}^{h} \frac{Debt_{t+j}^{R}}{GDP_{t-1+j}^{R}} \frac{GDP_{t-1+j}^{R}}{Trend_{t+j}^{GDPR}}}{\partial \sum_{j=0}^{h} g_{t+j}} \approx m_{h} \cdot \kappa_{h}$$
(B.V)

given Ass. I and Ass. II. Therefore, we adjust the debt multiplier following Eq. B.V. Note that the adjustment holds true also for the state-dependent case. Moreover, we adjust the standard error of the debt multiplier using standard delta methods.

APPENDIX C. DISTRIBUTION OF FISCAL SHOCKS ACROSS STATES

Figure 12 below reports the ratio between the Ramey instrument, used to identify exogenous fiscal expansions/contractions, and GDP, distinguishing between shocks in tight fiscal space (blue) and large fiscal space (blue). The picture shows that shocks are roughly balanced across regimes, in the sense that there is similar mass of fiscal policy shocks in periods of tight fiscal space and in periods of large fiscal space. Looking at the two biggest shocks recorded by Ramey, one is in a period of fiscal space, while the other coincides with the large fiscal space period. The graph also shows that shocks are balanced between negative and positive signs. Shocks are instead less balanced in terms of magnitude. However, this is a well-known characteristic regarding shocks identified Å la Ramey. Even in the linear analysis, WWII and the Korean war represent major fiscal policy shocks while the rest are much smaller in size. While we acknowledge this limitation, such problem is common to every paper employing the Ramey instrument, which is nonetheless one of the most common identification methods in fiscal policy.

FIGURE 12.





APPENDIX D. ADDITIONAL RESULTS ON THE FISCAL MULTIPLIER

D.1. **Federal Debt-to-GDP.** Tables 28 and 29 show the estimates for the fiscal multiplier according to the level of federal debt-to-GDP ratios. We define the state as *High Debt (Low Debt)* when the federal debt-to-GDP ratio is above (below) its historical median.⁵⁵ Consistently with Auerbach and Gorodnichenko (2017) and Huidrom et al. (2019), we find a multiplier that is higher in a low debt state. We do not find evidence to support a multiplier close to zero – or even negative – as in Ilzetzki et al. (2013). The multiplier is significantly different across the two states for most horizons when the shock series is instrumented with Ramey news; while we find no statistical significant difference using the Blanchard and Perotti shock (besides for the 5-years horizon multiplier).

⁵⁵The historical median for federal debt-to-GDP ratio is equal to 32.85%.

Horizon	Linear	Low Debt	High Debt	<i>p</i> -value Diff.
1-year	0.64555	1.96134	0.70015	0.32558
2	(0.15678)	(1.24905)	(0.19650)	
2-years	0.74378	1.58695	0.84027	0.35393
-	(0.15380)	(0.46191)	(0.19793)	
3-years	0.84977	1.93751	0.91943	0.00057***
-	(0.11385)	(0.19939)	(0.16250)	
4-years	0.87766	2.20669	0.90973	$6.03 \cdot 10^{-7***}$
-	(0.11009)	(0.20368)	(0.15726)	
5-years	0.98316	1.68699	1.04301	0.01065**
	(0.14459)	(0.15720)	(0.17652)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 28. Fiscal Space-dependent Fiscal Multiplier: Federal Debt-to-GDP - Ramey News Shock (1929-2015)

TABLE 29. Fiscal Space-dependent Fiscal Multiplier: Federal Debt-to-GDP - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Low Debt	High Debt	<i>p</i> -value Diff.
1-year	0.57740	0.03135	0.58395	0.47683
2	(0.07776)	(0.75732)	(0.08169)	
2-years	0.72720	0.62630	0.86743	0.57475
-	(0.10753)	(0.35351)	(0.12950)	
3-years	0.77148	1.22643	0.95096	0.26663
	(0.12386)	(0.16173)	(0.15893)	
4-years	0.82850	1.57550	1.01879	0.10046
	(0.15460)	(0.21285)	(0.22657)	
5-years	0.80963	1.50644	0.84209	0.01333**
	(0.19480)	(0.17706)	(0.18071)	
Signif. codes	0.01***	0.05**	0.1*	

D.2. Δ **Debt.** Tables 30 and 31 show the estimates for the fiscal multiplier according to the change in federal debt-to-GDP ratio. We define the state as *High* Δ *Debt* (*Low* Δ *Debt*) when the change in federal debt-to-GDP ratio is above (below) zero.⁵⁶ We find a multiplier that is higher in a decelerating debt state using Ramey news for the 4 and 5-years horizon. We find no statistical significant difference across states employing the Blanchard and Perotti shock.

⁵⁶This value represents also the historical median other than the turning point between accelerating and decelerating debt.

Horizon	Linear	Low Δ Debt	High Δ Debt	<i>p</i> -value Diff.
1-year	0.64555	0.75493	0.60510	0.62602
2	(0.15678)	(0.23394)	(0.17767)	
2-years	0.74378	0.93194	0.78165	0.47637
-	(0.15380)	(0.14717)	(0.14758)	
3-years	0.84977	1.08050	0.80710	0.10353
-	(0.11385)	(0.11204)	(0.12705)	
4-years	0.87766	1.15485	0.82131	0.06296*
	(0.11009)	(0.12138)	(0.13819)	
5-years	0.98316	1.31979	0.83600	0.04028**
	(0.14459)	(0.16888)	(0.16059)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 30. Fiscal Space-dependent Fiscal Multiplier: Change in Federal Debtto-GDP - Ramey News Shock (1929-2015)

TABLE 31. Fiscal Space-dependent Fiscal Multiplier: Change in Federal Debtto-GDP - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Low Δ Debt	High Δ Debt	<i>p</i> -value Diff.
1-year	0.57740	0.95312	0.61212	0.02633**
2	(0.07776)	(0.12153)	(0.05837)	
2-years	0.72720	1.08142	0.78136	0.12113
-	(0.10753)	(0.15878)	(0.09679)	
3-years	0.77148	1.13838	0.81918	0.11319
	(0.12386)	(0.16953)	(0.11446)	
4-years	0.82850	1.17187	0.86669	0.34797
	(0.15460)	(0.30239)	(0.13098)	
5-years	0.80963	2.90419	0.84928	0.93328
-	(0.19480)	(26.79590)	(0.13662)	
Signif. codes	0.01***	0.05**	0.1*	

D.3. Interaction of ZLB and Fiscal Space state. The two most remarkable periods of tight fiscal space are the Second World War and the Great financial Crisis jointly with its aftermath. These periods coincide with the most long-lasting periods under the zero lower bound. Thus, we estimate the multiplier depending on the interaction of the zero lower bound and tight fiscal space periods. We do so to see whether the presence of the zero lower bound biased our estimates. Tables 32 and 33 show no difference across states in the multiplier validating our baseline results.

Horizon	Linear	LFS & No ZLB	TFS & ZLB	<i>p</i> -value Diff.
1-year	0.64555	2.07650	1.01782	0.21670
2	(0.15678)	(0.86684)	(0.41416)	
2-years	0.74378	1.43534	0.55875	0.16245
-	(0.15380)	(0.46597)	(0.30145)	
3-years	0.84977	1.28415	0.74756	0.08978^{*}
-	(0.11385)	(0.24309)	(0.11227)	
4-years	0.87766	1.18503	0.11092	0.78251
-	(0.11009)	(0.21641)	(3.44917)	
5-years	0.98316	1.21257	0.96030	0.22542
-	(0.14459)	(0.20555)	(0.00684)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 32. Fiscal Space-dependent Fiscal Multiplier: Interaction ZLB and FS_1 - Ramey News Shock (1929-2015)

TABLE 33. Fiscal Space-dependent Fiscal Multiplier: Interaction ZLB and FS_1 - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	LFS & No ZLB	TFS & ZLB	<i>p</i> -value Diff.
1-year	0.57740	0.60104	0.37292	0.54738
2	(0.07776)	(0.25408)	(0.25914)	
2-years	0.72720	0.78203	4.78244	0.66873
-	(0.10753)	(0.19778)	(8.39845)	
3-years	0.77148	0.82977	1.64638	0.30290
	(0.12386)	(0.19360)	(0.78621)	
4-years	0.82850	0.81061	0.98918	0.53830
	(0.15460)	(0.20094)	(0.21637)	
5-years	0.80963	0.78085	2.73383	0.87672
	(0.19480)	(0.26161)	(12.56156)	
Signif. codes	0.01***	0.05**	0.1*	

D.4. **Slack.** We replicate the results in Ramey and Zubairy (2018) estimating the multiplier under recessions and expansions. Tables 34 and 35 show no statistical difference across states of slack and boom for the fiscal multiplier. This validates our empirical strategy and provides robustness for the results in Ramey and Zubairy (2018) given the different sample and the wider set of controls used in the present work.

Horizon	Linear	Expansion	Recession	<i>p</i> -value Diff.
1-year	0.64555	0.92530	-0.61172	0.00541***
2	(0.15678)	(0.25120)	(0.51275)	
2-years	0.74378	1.12149	-3.90960	0.03860**
-	(0.15380)	(0.27101)	(2.03599)	
3-years	0.84977	1.23876	-1.92392	0.44267
	(0.11385)	(0.21261)	(3.79865)	
4-years	0.87766	1.43659	1.92033	0.75045
	(0.11009)	(0.26078)	(1.54041)	
5-years	0.98316	1.65454	0.02609	0.59480
	(0.14459)	(0.25525)	(3.05506)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 34. Fiscal Space-dependent Fiscal Multiplier: Recession and Expansion^{*a*} - Ramey News Shock (1929-2015)

^{*a*}Unemployment > 8%

TABLE 35. Fiscal Space-dependent Fiscal Multiplier: Recession and Expansion - Blanchard and Perotti Shock (1929-2015)

Horizon	Linear	Expansion	Recession	<i>p</i> -value Diff.
1-year	0.57740	0.66432	0.69821	0.96562
2	(0.07776)	(0.08819)	(0.65954)	
2-years	0.72720	0.77796	-1.87081	0.31921
-	(0.10753)	(0.14893)	(2.56745)	
3-years	0.77148	0.90941	0.89519	0.96985
-	(0.12386)	(0.21230)	(0.31763)	
4-years	0.82850	1.03321	0.90694	0.72403
-	(0.15460)	(0.29877)	(0.21281)	
5-years	0.80963	0.93673	0.86724	0.85744
-	(0.19480)	(0.37941)	(0.07089)	
Signif. codes	0.01***	0.05**	0.1*	

TABLE 36. Bootstrapped correlations (95 % confidence interval) among Fiscal space, potential CBO output, unemployment rate, federal debt-to-GDP, change in debt, growth rate of debt series.

	Median	Lower bound	Upper bound
$\mathbb{C}orr(FS1_t, \tilde{y}_t)$	0.396	0.304	0.482
$\mathbb{C}orr(FS2_t, \tilde{y}_t)$	0.002	-0.112	0.112
$\mathbb{C}orr(FS3_t, \tilde{y}_t)$	-0.347	-0.416	-0.279
$\mathbb{C}orr(FS4_t, \tilde{y}_t)$	-0.002	-0.078	0.072
$Corr(FS1_t, U_t)$	0.079	-0.0003	0.158
$\mathbb{C}orr(FS2_t, U_t)$	-0.042	-0.141	0.055
$Corr(FS3_t, U_t)$	0.685	0.605	0.770
$\mathbb{C}orr(FS4_t, U_t)$	0.427	0.269	0.579
$\mathbb{C}orr(FS1_t, \frac{B}{GDP_t})$	0.413	0.296	0.577
$\mathbb{C}orr(FS2_t, \frac{B}{GDP_t})$	0.244	0.090	0.395
$\mathbb{C}orr(FS3_t, \frac{\overline{B}}{\overline{GDP}t})$	-0.084	-0.145	-0.026
$\mathbb{C}orr(FS4_t, \frac{B}{GDP_t})$	0.131	0.039	0.222
$Corr(FS1_t, \Delta B_t)$	0.394	0.312	0.470
$Corr(FS2_t, \Delta B_t)$	0.242	0.140	0.337
$\mathbb{C}orr(FS3_t, \Delta B_t)$	-0.055	-0.102	-0.009
$\mathbb{C}orr(FS4_t, \Delta B_t)$	0.117	0.053	0.176

The Table shows the estimates for median non-parametric bootstrapped correlation coefficients and their intervals at the 95% confidence level. Intervals are calculated using the normal approximation. FSj_t (j = 1, 2, 3, 4) are the fiscal space proxies, \tilde{y}_t is the potential output (latest estimates from CBO), U_t is the unemployment rate, $\frac{B}{GDP_t}$ is the federal debt-to-GDP ratio and ΔB_t represents its change.



FIGURE 13. *FS* dummies and NBER recession dates.

This figure shows the periods of Tight Fiscal Space via our *FS* state dummies and of Recessions (identifying recessions with NBER recession dates). See also Table 2 for more details.





The Figure shows the estimates for the impulse response function of real government spending scaled by trend output (*G*) and real GDP scaled by trend output (*GDP*) following a Blanchard and Perotti shock.



FIGURE 15. *C* and *I* IRFs - *FS*₁ - Blanchard and Perotti (1929-2015)

The Figure shows the estimates for the impulse response function of real consumption scaled by trend output (C) and real total private investment scaled by trend output (I) following a Blanchard and Perotti shock.



FIGURE 16. Cons. and Inv. multipliers - FS₁ - Blanchard and Perotti (1929-2015)





The Figure shows the estimates for the integral multiplier of federal debt. The correct multiplier is derived by adjusting the resulting 2SLS, see Appendix B for details.



FIGURE 18. AMTR, P. Debt, deficit IRFs - FS₁ - Blanchard and Perotti (1929-2015)

The Figure shows the estimates for the impulse response function of the average marginal tax rate as in Barro and Redlick (2011) and Bernardini and Peersman (2018) (*AMTR*), federal debt-to-lagged GDP, deficit-to-GDP following a Blanchard and Perotti shock.



FIGURE 19. log(IPGDP), Y10, C.B. spread IRFs - FS_1 - Blanchard and Perotti (1929-2015)



The Figure shows the estimates for the impulse response function of the logarithm of the implicit price deflator (log(IPGDP)), 10-years government bond yield (Y10), corporate bond spread (C.B. spread) following a Blanchard and Perotti shock.

FIGURE 20. (1 year average) FS_1 (in real terms) IRFs - Ramey news and Blanchard & Perotti shock (1929-2015).



Ramey News - (1y avg) FS_1 (L/T FS)

Blanchard & Perotti - (1y avg) FS_1 (L/T FS)



The Figure shows the estimates for the impulse response function of the Fiscal Space (1 year average) series (in real terms) following both a Ramey news and Blanchard & Perotti shock.

RECENTLY PUBLISHED "TEMI" (*)

- N. 1271 An analysis of sovereign credit risk premia in the euro area: are they explained by local or global factors?, by Sara Cecchetti (March 2020).
- N. 1272 Mutual funds' performance: the role of distribution networks and bank affiliation, by Giorgio Albareto, Andrea Cardillo, Andrea Hamaui and Giuseppe Marinelli (April 2020).
- N. 1273 Immigration and the fear of unemployment: evidence from individual perceptions in Italy, by Eleonora Porreca and Alfonso Rosolia (April 2020).
- N. 1274 Bridge Proxy-SVAR: estimating the macroeconomic effects of shocks identified at high-frequency, by Andrea Gazzani and Alejandro Vicondoa (April 2020).
- N. 1275 Monetary policy gradualism and the nonlinear effects of monetary shocks, by Luca Metelli, Filippo Natoli and Luca Rossi (April 2020).
- N. 1276 Spend today or spend tomorrow? The role of inflation expectations in consumer behaviour, by Concetta Rondinelli and Roberta Zizza (April 2020).
- N. 1277 Going the extra mile: effort by workers and job-seekers, by Matthias S. Hertweck, Vivien Lewis and Stefania Villa (June 2020).
- N. 1278 *Trainspotting: board appointments in private firms*, by Audinga Baltrunaite and Egle Karmaziene (June 2020).
- N. 1279 *The role of bank supply in the Italian credit market: evidence from a new regional survey*, by Andrea Orame (June 2020).
- N. 1280 The non-linear effects of the Fed asset purchases, by Alessio Anzuini (June 2020).
- N. 1281 The effects of shop opening hours deregulation: evidence from Italy, by Lucia Rizzica, Giacomo Roma and Gabriele Rovigatti (June 2020).
- N. 1282 *How do house prices respond to mortgage supply*?, by Guglielmo Barone, Francesco David, Guido de Blasio and Sauro Mocetti (June 2020).
- N. 1283 The macroeconomics of hedging income shares, by Adriana Grasso, Juan Passadore and Facundo Piguillem (June 2020).
- N. 1284 Uncertainty matters: evidence from a high-frequency identification strategy, by Piergiorgio Alessandri, Andrea Gazzani and Alejandro Vicondoa (June 2020).
- N. 1285 *The economic drivers of volatility and uncertainty*, by Andrea Carriero, Francesco Corsello and Massimiliano Marcellino (July 2020).
- N. 1286 *Protectionism and the effective lower bound in the euro area*, by Pietro Cova, Alessandro Notarpietro and Massimiliano Pisani (July 2020).
- N. 1287 *Court efficiency and aggregate productivity: the credit channel*, by Guzmán González-Torres and Giacomo Rodano (July 2020).
- N. 1288 *The time-varying risk of Italian GDP*, by Fabio Busetti, Michele Caivano, Davide Delle Monache and Claudia Pacella (July 2020).
- N. 1289 *Trade shocks and credit reallocation*, by Stefano Federico, Fadi Hassan and Veronica Rappoport (July 2020).
- N. 1290 The impact of the IRB approach on the relationship between the cost of credit for public companies and financial market conditions, by Raffaele Gallo (July 2020).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2018

- ACCETTURO A., V. DI GIACINTO, G. MICUCCI and M. PAGNINI, Geography, productivity and trade: does selection explain why some locations are more productive than others?, Journal of Regional Science, v. 58, 5, pp. 949-979, WP 910 (April 2013).
- ADAMOPOULOU A. and E. KAYA, *Young adults living with their parents and the influence of peers*, Oxford Bulletin of Economics and Statistics, v. 80, pp. 689-713, WP 1038 (November 2015).
- ANDINI M., E. CIANI, G. DE BLASIO, A. D'IGNAZIO and V. SILVESTRINI, *Targeting with machine learning:* an application to a tax rebate program in Italy, Journal of Economic Behavior & Organization, v. 156, pp. 86-102, WP 1158 (December 2017).
- BARONE G., G. DE BLASIO and S. MOCETTI, *The real effects of credit crunch in the great recession: evidence from Italian provinces*, Regional Science and Urban Economics, v. 70, pp. 352-59, **WP 1057 (March 2016)**.
- BELOTTI F. and G. ILARDI Consistent inference in fixed-effects stochastic frontier models, Journal of Econometrics, v. 202, 2, pp. 161-177, WP 1147 (October 2017).
- BERTON F., S. MOCETTI, A. PRESBITERO and M. RICHIARDI, *Banks, firms, and jobs,* Review of Financial Studies, v.31, 6, pp. 2113-2156, WP 1097 (February 2017).
- BOFONDI M., L. CARPINELLI and E. SETTE, *Credit supply during a sovereign debt crisis*, Journal of the European Economic Association, v.16, 3, pp. 696-729, WP 909 (April 2013).
- BOKAN N., A. GERALI, S. GOMES, P. JACQUINOT and M. PISANI, EAGLE-FLI: a macroeconomic model of banking and financial interdependence in the euro area, Economic Modelling, v. 69, C, pp. 249-280, WP 1064 (April 2016).
- BRILLI Y. and M. TONELLO, Does increasing compulsory education reduce or displace adolescent crime? New evidence from administrative and victimization data, CESifo Economic Studies, v. 64, 1, pp. 15–4, WP 1008 (April 2015).
- BUONO I. and S. FORMAI *The heterogeneous response of domestic sales and exports to bank credit shocks,* Journal of International Economics, v. 113, pp. 55-73, WP 1066 (March 2018).
- BURLON L., A. GERALI, A. NOTARPIETRO and M. PISANI, Non-standard monetary policy, asset prices and macroprudential policy in a monetary union, Journal of International Money and Finance, v. 88, pp. 25-53, WP 1089 (October 2016).
- CARTA F. and M. DE PHLIPPIS, You've Come a long way, baby. Husbands' commuting time and family labour supply, Regional Science and Urban Economics, v. 69, pp. 25-37, WP 1003 (March 2015).
- CARTA F. and L. RIZZICA, *Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy*, Journal of Public Economics, v. 158, pp. 79-102, WP 1030 (October 2015).
- CASIRAGHI M., E. GAIOTTI, L. RODANO and A. SECCHI, A "Reverse Robin Hood"? The distributional implications of non-standard monetary policy for Italian households, Journal of International Money and Finance, v. 85, pp. 215-235, WP 1077 (July 2016).
- CIANI E. and C. DEIANA, *No Free lunch, buddy: housing transfers and informal care later in life*, Review of Economics of the Household, v.16, 4, pp. 971-1001, **WP 1117 (June 2017).**
- CIPRIANI M., A. GUARINO, G. GUAZZAROTTI, F. TAGLIATI and S. FISHER, *Informational contagion in the laboratory*, Review of Finance, v. 22, 3, pp. 877-904, WP 1063 (April 2016).
- DE BLASIO G, S. DE MITRI, S. D'IGNAZIO, P. FINALDI RUSSO and L. STOPPANI, *Public guarantees to SME borrowing*. A RDD evaluation, Journal of Banking & Finance, v. 96, pp. 73-86, WP 1111 (April 2017).
- GERALI A., A. LOCARNO, A. NOTARPIETRO and M. PISANI, *The sovereign crisis and Italy's potential output*, Journal of Policy Modeling, v. 40, 2, pp. 418-433, **WP 1010 (June 2015).**
- LIBERATI D., An estimated DSGE model with search and matching frictions in the credit market, International Journal of Monetary Economics and Finance (IJMEF), v. 11, 6, pp. 567-617, WP 986 (November 2014).
- LINARELLO A., Direct and indirect effects of trade liberalization: evidence from Chile, Journal of Development Economics, v. 134, pp. 160-175, WP 994 (December 2014).
- NATOLI F. and L. SIGALOTTI, *Tail co-movement in inflation expectations as an indicator of anchoring,* International Journal of Central Banking, v. 14, 1, pp. 35-71, WP 1025 (July 2015).
- NUCCI F. and M. RIGGI, *Labor force participation, wage rigidities, and inflation,* Journal of Macroeconomics, v. 55, 3 pp. 274-292, WP 1054 (March 2016).
- RIGON M. and F. ZANETTI, *Optimal monetary policy and fiscal policy interaction in a non_ricardian economy,* International Journal of Central Banking, v. 14 3, pp. 389-436, WP 1155 (December 2017).

SEGURA A., Why did sponsor banks rescue their SIVs?, Review of Finance, v. 22, 2, pp. 661-697, WP 1100 (February 2017).

2019

- ALBANESE G., M. CIOFFI and P. TOMMASINO, *Legislators' behaviour and electoral rules: evidence from an Italian reform*, European Journal of Political Economy, v. 59, pp. 423-444, **WP 1135 (September 2017).**
- APRIGLIANO V., G. ARDIZZI and L. MONTEFORTE, Using the payment system data to forecast the economic activity, International Journal of Central Banking, v. 15, 4, pp. 55-80, WP 1098 (February 2017).
- ARNAUDO D., G. MICUCCI, M. RIGON and P. ROSSI, Should I stay or should I go? Firms' mobility across banks in the aftermath of the financial crisis, Italian Economic Journal / Rivista italiana degli economisti, v. 5, 1, pp. 17-37, WP 1086 (October 2016).
- BASSO G., F. D'AMURI and G. PERI, *Immigrants, labor market dynamics and adjustment to shocks in the euro area,* IMF Economic Review, v. 67, 3, pp. 528-572, WP 1195 (November 2018).
- BATINI N., G. MELINA and S. VILLA, *Fiscal buffers, private debt, and recession: the good, the bad and the ugly,* Journal of Macroeconomics, v. 62, WP 1186 (July 2018).
- BURLON L., A. NOTARPIETRO and M. PISANI, *Macroeconomic effects of an open-ended asset purchase programme*, Journal of Policy Modeling, v. 41, 6, pp. 1144-1159, **WP 1185 (July 2018).**
- BUSETTI F. and M. CAIVANO, Low frequency drivers of the real interest rate: empirical evidence for advanced economies, International Finance, v. 22, 2, pp. 171-185, WP 1132 (September 2017).
- CAPPELLETTI G., G. GUAZZAROTTI and P. TOMMASINO, *Tax deferral and mutual fund inflows: evidence from a quasi-natural experiment*, Fiscal Studies, v. 40, 2, pp. 211-237, **WP 938 (November 2013).**
- CARDANI R., A. PACCAGNINI and S. VILLA, Forecasting with instabilities: an application to DSGE models with financial frictions, Journal of Macroeconomics, v. 61, WP 1234 (September 2019).
- CHIADES P., L. GRECO, V. MENGOTTO, L. MORETTI and P. VALBONESI, Fiscal consolidation by intergovernmental transfers cuts? The unpleasant effect on expenditure arrears, Economic Modelling, v. 77, pp. 266-275, WP 985 (July 2016).
- CIANI E., F. DAVID and G. DE BLASIO, *Local responses to labor demand shocks: a re-assessment of the case of Italy*, Regional Science and Urban Economics, v. 75, pp. 1-21, WP 1112 (April 2017).
- CIANI E. and P. FISHER, *Dif-in-dif estimators of multiplicative treatment effects*, Journal of Econometric Methods, v. 8. 1, pp. 1-10, WP 985 (November 2014).
- CIAPANNA E. and M. TABOGA, *Bayesian analysis of coefficient instability in dynamic regressions*, Econometrics, MDPI, Open Access Journal, v. 7, 3, pp.1-32, WP 836 (November 2011).
- COLETTA M., R. DE BONIS and S. PIERMATTEI, *Household debt in OECD countries: the role of supply-side* and demand-side factors, Social Indicators Research, v. 143, 3, pp. 1185–1217, **WP 989 (November** 2014).
- COVA P., P. PAGANO and M. PISANI, *Domestic and international effects of the Eurosystem Expanded Asset Purchase Programme*, IMF Economic Review, v. 67, 2, pp. 315-348, WP 1036 (October 2015).
- ERCOLANI V. and J. VALLE E AZEVEDO, *How can the government spending multiplier be small at the zero lower bound?*, Macroeconomic Dynamics, v. 23, 8. pp. 3457-2482, **WP 1174 (April 2018).**
- FERRERO G., M. GROSS and S. NERI, *On secular stagnation and low interest rates: demography matters,* International Finance, v. 22, 3, pp. 262-278, **WP 1137 (September 2017).**
- FOA G., L. GAMBACORTA, L. GUISO and P. E. MISTRULLI, *The supply side of household finance*, Review of Financial Studies, v.32, 10, pp. 3762-3798, **WP 1044 (November 2015).**
- GIORDANO C., M. MARINUCCI and A. SILVESTRINI, *The macro determinants of firms' and households' investment: evidence from Italy*, Economic Modelling, v. 78, pp. 118-133, WP 1167 (March 2018).
- GOMELLINI M., D. PELLEGRINO and F. GIFFONI, *Human capital and urban growth in Italy*,1981-2001, Review of Urban & Regional Development Studies, v. 31, 2, pp. 77-101, **WP 1127 (July 2017).**
- MAGRI S., Are lenders using risk-based pricing in the Italian consumer loan market? The effect of the 2008 crisis, Journal of Credit Risk, v. 15, 1, pp. 27-65, WP 1164 (January 2018).
- MAKINEN T., A. MERCATANTI and A. SILVESTRINI, *The role of financial factors for european corporate investment*, Journal of International Money and Finance, v. 96, pp. 246-258, **WP 1148 (October 2017).**
- MIGLIETTA A., C. PICILLO and M. PIETRUNTI, *The impact of margin policies on the Italian repo market*, The North American Journal of Economics and Finance, v. 50, **WP 1028 (October 2015).**

- MONTEFORTE L. and V. RAPONI, Short-term forecasts of economic activity: are fortnightly factors useful?, Journal of Forecasting, v. 38, 3, pp. 207-221, WP 1177 (June 2018).
- NERI S. and A. NOTARPIETRO, Collateral constraints, the zero lower bound, and the debt-deflation mechanism, Economics Letters, v. 174, pp. 144-148, WP 1040 (November 2015).
- PEREDA FERNANDEZ S., *Teachers and cheaters. Just an anagram?*, Journal of Human Capital, v. 13, 4, pp. 635-669, WP 1047 (January 2016).
- RIGGI M., Capital destruction, jobless recoveries, and the discipline device role of unemployment, Macroeconomic Dynamics, v. 23, 2, pp. 590-624, WP 871 (July 2012).

2020

- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, WP 1110 (April 2017).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, **WP 1059 (March 2016).**
- RIZZICA L., Raising aspirations and higher education. Evidence from the UK's widening participation policy, Journal of Labor Economics, v. 38, 1, pp. 183-214, WP 1188 (September 2018).

FORTHCOMING

- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, **WP 974 (October 2014).**
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, *Discretion and supplier selection in public procurement*, Journal of Law, Economics, and Organization, WP 1178 (June 2018).
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, WP 1201 (November 2018).
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy,* Journal of International Economics, **WP 1032 (October 2015).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome*, Journal of Regional Science, WP 1208 (February 2019).
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, **WP 1152 (December 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, WP 1156 (December 2017).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises*, Review of World Economics, WP 1217 (April 2019).
- GERALI A. and S. NERI, *Natural rates across the Atlantic*, Journal of Macroeconomics, WP 1140 (September 2017).
- LIBERATI D. and M. LOBERTO, *Taxation and housing markets with search frictions*, Journal of Housing Economics, WP 1105 (March 2017).
- LOSCHIAVO D., Household debt and income inequality: evidence from Italian survey data, Review of Income and Wealth, WP 1095 (January 2017).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility*, Journal of Human Resources, WP 1182 (July 2018).

- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, **WP 1200** (November 2018).
- PANCRAZI R. and M. PIETRUNTI, *Natural expectations and home equity extraction*, Journal of Housing Economics, **WP 984 (November 2014).**
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, WP 1092 (January 2017).

RAINONE E., The network nature of otc interest rates, Journal of Financial Markets, WP 1022 (July 2015).

- SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, WP 1146 (October 2017).
- SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, Bank Quality, judicial efficiency and loan repayment delays in Italy, Journal of Finance, WP 1072 (July 2016).